INTEGRATED PROJECT DELIVERY AGREEMENT

BUSINESS TERMS SHEET

The key business terms of this Agreement are set forth below and included in the Agreement:

Senior Management Team (SMT)	
Project Management Team (PMT)	
Owner's Budget	
Estimated Final Cost	
Target Cost	
Added Value Incentive Items	
ICL amounts & ICL Percentages of Risk/Reward Team	
Prime Consultant's Overhead Multiplier (Exhibit E, Section 1.1)	
General Contractor's Overhead	
Change Order % (ICL Change Limitation Section 12.4.)	
Date for Substantial Performance of the Project	
Date for Final Completion of the Project	

EXHIBITS

Exhibit		Status at Agreement Signing
Exhibit A	Definitions	Included
A-1	Town Standards	Included
Exhibit B	Project Objective	Included
B-1	Owner's Program Requirements	Included
B-1a	Fire Hall Program Summary	Included
B-1b	Oakville Arena Program Summary	Included
B-2	Target Cost Breakdown	Included
B-2a	Risk and Innovations	Included
B-3	Added Value Incentive Items	By Amendment
B-4	Milestones Definition	Included
B-5	Implementation Document List	By Amendment
Exhibit C	Contract Task Matrix	Included
C-1	RASCI Matrix Definitions	Included
Exhibit D	Intentionally Omitted	
Exhibit E	Prime Consultant's Chargeable Costs	Included
E-1	Billable Rates	Included
Exhibit F	General Contractor's Chargeable Costs	Included
F-1	Billable Rates	Included
F-2	Equipment Rates	By Amendment
1 2	Equipment Nates	by / anonamone
Exhibit G	ICL Distribution Spreadsheet	Included
Exhibit H	Project Insurance Requirements	Included
LAHIDILH	1 Tojout insulance requirements	molaucu
Exhibit I	Key Employees	Included

Items shown as Included as of the signing of the Agreement are subject to amendment in accordance with the terms of the Agreement.

TABLE OF CONTENTS

			Page
1.	DEFI	NITIONS	2
	1.1	Defined Terms	2
2.	THE	PROJECT	2
	2.1	The Project	2
	2.1	Project Objective	
	2.3	Project Phases and Stages	
	2.4	Collaboration	
3.	IPD 1	ГЕАМ	3
	3.1	Forming the IPD Team	3
	3.2	IPD Team Purpose and Intent.	
	3.3	IPD Team Alignment.	
	3.4	Trust	
	3.5	Relationship of the IPD Team Members	
4.	THE	PARTIES	4
	4.1	Relationship of the Parties	4
	4.2	Key Employees	
	4.3	Owner	
	4.4	Prime Consultant	5
	4.5	General Contractor	7
5.	PRO	JECT MANAGEMENT	9
	5.1	Project Management Team	9
	5.2	Senior Management Team; Owner's Directive	
	5.3	Written Confirmation of Decisions	
	5.4	Personnel Management	11
6.	SUB	CONTRACTS AND CONSULTING AGREEMENTS	12
	6.1	Written Agreements	12
	6.2	Licensing Requirements	
	6.3	Required Pass Through Provisions	
	6.4	Separate Contractors	14
7.	RESI	PONSIBILITIES BY PHASE	14
	7.1	Validation Phase	
	7.2	Design/Preconstruction Phase	
	7.3	Construction Phase	18
8	COM	IPENSATION	19

	8.1	General	19
	8.2	General Contractor Self-Performed Work	19
	8.3	Incentive Compensation Layer	20
9.	RISK/	REWARD PROGRAM	20
0.	racia	TEW III TROOF III	20
	9.1	General	
	9.2	General Contractor's ICL Pass-through	
	9.3	Prime Consultant's ICL Pass-through	
	9.4	ICL Distribution Schedule	
	9.5	Milestone ICL Payments	
	9.6	Milestone Estimate	
	9.7	Milestone Overpayment	
	9.8	Design/Preconstruction Phase ICL Adjustments	
	9.9	Construction Phase ICL Adjustments	23
10.	PAYM	1ENT	23
	10.1	Chargeable Costs	23
	10.1	Periodic Payments	
	10.2	Final Payment	
	10.3	Materials and Equipment	
	10.4	Supporting Documents	
	10.5	Right to Withhold	
	10.0		
	10.7	No Right to Stop WorkReliance	
	10.8		
	10.9	Warranty of Title No Waiver	
	10.10		
	10.11	Payment to Subcontractors and Suppliers	
	10.12	,	
11.	CONT	FRACT TIME	27
	11.1	Contract Time	
	11.2	Commencement of the Construction Work	
	11.3	Project Staging.	
	11.4	Project Scheduling	27
	11.5	Permitted Delays	
	11.6	Notice	
	11.7	Schedule Slippage	
	11.8	Acceleration	28
12.	CHAN	IGES	28
	12.1	Permitted Changes	28
	12.2	Changes to Cost	29
	12.3	Changes to the Contract Time	
	12.4	Limitation on Changes to ICL	
	12.5	Change Order Procedure	
	12.6	Owner's Directive	

13.	LIABILITY ALLOCATION			
	13.1	Waiver of Liability	29	
	13.2	Allowed Claims	30	
	13.3	Indemnification	31	
	13.4	Site Safety	33	
	13.5	Joint Defense of Third-Party Claims	34	
14.	PROJ	ECT INSURANCE	34	
	14.1	General	34	
	14.2	Bonding or Subcontractor Default Insurance	34	
15.	WARF	RANTY	34	
	15.1	Warranty and Correction of Work	34	
16.	DISPL	JTE RESOLUTION	35	
	16.1	Scope	35	
	16.2	Continued Performance	35	
	16.3	Permitted Disputes	35	
	16.4	Notice	35	
	16.5	Senior Management Team Meeting	35	
	16.6	Mediation	35	
	16.7	Litigation	36	
17.	DEFA	ULT, SUSPENSION AND TERMINATION	36	
	17.1	Termination for Convenience		
	17.2	Suspension		
	17.3	Owner Termination for Cause		
	17.4	Prime Consultant/General Contractor Termination for Cause	38	
18.	MISCE	ELLANEOUS PROVISIONS	38	
	18.1	License		
	18.2	Confidentiality		
	18.3	Compliance With Anti-Corruption Laws		
	18.4	Notices		
	18.5	Governing Law		
	18.6	Commencement of Statute of Limitations		
	18.7	Assignment		
	18.8	Notice of Dissolution and Other Change of Business		
	18.9	Severability		
	18.10	,		
	18.11	Rights and Remedies		
	18.12			
	18.13			
	18.1 4	Execution	41	

18.15	Counterparts	.41	
18.16	Exhibits	.41	
18.17	Entire Agreement	.41	

INTEGRATED PROJECT DELIVERY AGREEMENT

-	This Integrated Pro	oject Delivery	Agreement (("Agreement") is entered i	nto on [inse	rt date],	2016
(("Effective Date")) among:						

The Corporation of the Town of Oakville ("Owner"):

Diamond and Schmitt Architects Incorporated ("Prime Consultant")

Graham Construction and Engineering LP ("General Contractor")

1. **DEFINITIONS**

1.1 Defined Terms. Defined terms and titles of Exhibits will be capitalized throughout the Agreement. The definitions for this Agreement are set forth in alphabetical order in Exhibit A. The Owner, Prime Consultant, and General Contractor may be individually referred to as a Party and will be collectively referred to throughout this Agreement as the Parties. References to "subcontractor" or "subconsultant" in lower case refer to all subcontractors and subconsultants of any tier, including the Risk/Reward subcontractors and Risk/Reward subconsultants. "Article" refers to the complete text under a first level heading, such as "2. The Project." "Section" refers to a specific numbered paragraph, such as "2.2 Project Objective."

2. THE PROJECT

2.1 The Project. The Project includes the conversion of the Oakville Arena into a community centre, the construction of a new Fire Hall and all associated site works. The Project will convert the Oakville Arena to an approximately 65,000 square feet facility with new steel framed roof structure constructed over the existing Hipel truss system, maintaining this heritage feature. The Project is to include an NHL size ice pad, a two bay garage, a senior centre with amenities, public meeting spaces, a full size gymnasium, interior walking track, and exterior washroom facilities to be used by the public visiting the sports field. The Project will also include the construction of a new Fire Hall on the corner of Kerr and Rebecca Streets. At a minimum, all existing park amenities are to be maintained and parking and vehicular access maximized for operational needs.

The Project must meet all applicable Town standards and All Applicable Laws.

- **2.2 Project Objective.** The Parties will jointly develop the Project Objective based upon the Owner's requirements, goals, and constraints. The Project Objective is comprised of the Base Program, Target Cost, Added Value Incentive Items, Implementation Documents, and Contract Time, and any other objectives agreed by the Parties. The Project Objective establishes the Project requirements and standards for measuring the Project's success. The various components of the Project Objective may be incorporated into the Agreement through Amendment upon recommendation of the Project Management Team and approval of the Senior Management Team.
- **2.3 Project Phases and Stages.** This Project has 3 phases of Work: Validation Phase, Design/Preconstruction Phase, and Construction Phase. The Project may also be divided into one or more Project Stages (eg., the Fire Hall may be one Project Stage and the Oakville Arena may be a different Project Stage). At any point in time, Work in different Project Stages may be in different Project Phases. For example, Work may be in the Construction Phase for one Project Stage while in the Design/Preconstruction Phase of a different Project Stage.
- **2.4 Collaboration.** The Owner, Prime Consultant, and General Contractor will work together to achieve the Project Objective by:
- 2.4.1 individually performing the Contract Tasks designated as their respective responsibilities in the Contract Task Matrix (Exhibit C);

- 2.4.2 jointly managing the Project through the PMT pursuant to Article 4;
- 2.4.3 cooperating with and assisting the PMT and the Project Implementation Teams ("PITS"), subject to the limits of their respective professional expertise, licensing, and abilities:
- 2.4.4 executing the Work according to Lean principles as described in the Validation Report;
- 2.4.5 fulfilling their obligations as IPD Team Members as described in Article 3; and
- 2.4.6 complying with all other obligations, terms, and conditions of the Contract Documents.

3. IPD TEAM

- **3.1 Forming the IPD Team.** Prime Consultant, Owner, and General Contractor agree to form an integrated team to facilitate the planning, designing, constructing, and commissioning of the Project. This IPD Team is comprised of (i) the Prime Consultant and the Prime Consultant's Subconsultants, (ii) General Contractor, Subcontractors, and Suppliers, and (iii) Owner and its Separate Contractors (collectively, the "Integrated Project Delivery Team" or "IPD Team"; and individually, each an "IPD Team Member"). The IPD Team Members will make reasonable commercial efforts to share relevant information in a timely manner and to cooperatively collaborate for the benefit of the Project. Each of the IPD Team Members will be required to sign a Joining Agreement that signifies its support for Lean and integrated project delivery principles and procedures.
- **3.2 IPD Team Purpose and Intent.** The purpose of the IPD Team is to facilitate collaborative planning, design, construction, and commissioning of the Project. In order to permit the IPD Team Members to gain the benefit of an open and creative learning environment, each of the IPD Team Members shall make reasonable commercial efforts to:
- 3.2.1 share relevant ideas freely and in a timely manner and to establish an atmosphere of mutual respect and tolerance;
- 3.2.2 work together and individually to achieve transparent and cooperative exchange of relevant information relating to the Project, and to share ideas for improving Project delivery;
- 3.2.3 actively promote harmony, collaboration, and cooperation among all entities performing on the Project; and
- 3.2.4 provide both traditional and exceptionally collaborative reconstruction services throughout the Validation Phase and early in the Design/Preconstruction Phase to facilitate an integrated, collaborative design process.
- **3.3 IPD Team Alignment.** The integrated project delivery approach recognizes that each IPD Team Member's success is tied directly to the success of all other members of the IPD Team and encourages and requires the IPD Team Members to organize and integrate their

respective roles, responsibilities, and expertise, to identify and align their respective expectations and objectives, to commit to open communications, transparent decision-making, proactive interaction, problem-solving, the sharing of ideas, to continuously seek to improve the Project planning, design and construction processes, and, as it relates to Risk/Reward Team Members, to share both the risks and rewards associated with achieving the Project objectives.

- 3.4 Trust. The IPD Team Members accept the relationship of mutual trust and confidence established with each other by this Agreement, and promise to furnish their professional skill and judgment and to collaborate and cooperate with each other and with other IPD Team Members in actively pursuing an integrated project and furthering the interests of the Project. The IPD Team Members recognize that each of their opportunities to succeed on the Project is directly tied to the performance of other IPD Team Members. The IPD Team Members agree to work together in the spirit of cooperation, collaboration, and mutual respect for the benefit of the Project, and within the limits of their professional expertise, licensing, and abilities. Throughout the Project, the IPD Team Members shall use their best efforts to perform, or cause to be performed, the Work and the Design Services in an expeditious and economical manner consistent with the interests of the Project. To the extent directed by the PMT, the Parties will utilize LeanTM principles and techniques as developed or defined by the Lean Construction InstituteTM.
- 3.5 Relationship of the IPD Team Members. Notwithstanding the commitments in this Article 3 and elsewhere in this Agreement, no fiduciary relationship is intended and nothing in this Agreement shall create, or be deemed or implied to create, a fiduciary relationship between or among the Parties, the IPD Team, or any other IPD Team Members.

4. THE PARTIES

- Relationship of the Parties. Although this Agreement establishes a relationship of mutual trust and good faith among the Parties, who recognize that their individual success is directly tied to the performance of other IPD Team Members, it does not create an agency relationship, fiduciary relationship, partnership, or joint venture among or between the Parties. The Prime Consultant and General Contractor are each independent contractors solely responsible for directing and managing their own forces and services within their respective area of responsibility as described in Sections 4.4 through 4.5. The Parties acknowledge that this Agreement is not a design-build agreement and that each Party, and each individual entity that is a Party, is responsible for its own errors, omissions, or construction defects to the extent provided in this Agreement. Likewise, nothing contained in this Agreement makes any Party jointly and severally liable for the negligent acts, omissions, or breaches of this Agreement of any other Party, except that (a) the General Contractor is responsible for the acts, errors, and omissions of its subcontractors, and the Prime Consultant is responsible for the acts, errors, and omissions of its subconsultants; and (b) the Incentive Compensation Layer may be eroded if errors or omissions of Prime Consultant, General Contractor, or those for whom they are responsible increase the incurred Chargeable Costs.
- **4.2 Key Employees.** Prime Consultant and General Contractor acknowledge that in entering into this Agreement, Owner has relied upon each assigning to this Project the key employees listed in Exhibit I, each of which must perform their respective work or services throughout the term of this Agreement. No key employee may be replaced, reassigned, or removed without Owner's written approval, which will not be unreasonably withheld. Owner may reasonably require replacement of any key employee. Replacement key employees,

whether replacement is required by Owner or agreed to by Owner, are key employees that may not be replaced, reassigned or removed except as allowed in this Section 4.2.

- **4.3 Owner.** The Owner will actively participate in the Target Value Design process and one or more representatives of the Owner serve as a member(s) of the Senior Management Team and the Project Management Team. The Owner is responsible for providing the Owner's requirements, goals, and limitations that must be accommodated in the Project and will actively participate in developing and documenting the Project Objective.
- 4.3.1 Owner Provided Information. Owner will provide a legal description of the property, access to all existing documentation, and all geotechnical or environmental impact reports, surveys, and other reports as may be required for proper performance of the Work and that are in Owner's possession. Risk/Reward Team members may not rely upon the completeness and accuracy of the information provided by the Owner at the time of the RFP. The PMT may decide that further studies are required. The costs of such further studies will be Chargeable Costs.
- 4.3.2 Permits and Fees. The Owner will pay for all entitlements, easements, assessments, and fees required for the development, use, or occupancy of the Project. Permits and fees related to the Construction Work are a Chargeable Cost. The General Contractor will apply for Permits (s. 4.5.4 h)).
- 4.3.3 Testing and Inspections. Owner will be responsible for all third party testing, inspections, or commissioning. General Contractor will be responsible for any other inspections and reports required by law or by the Contract Documents. Test and inspection costs are Chargeable Costs. However, subcontractors will bear all costs associated with reinspection if the re-inspection is caused by the negligence of the subcontractor or because the subcontractor's Construction Work was not ready for inspection. Non-conforming Construction Work that was performed or installed by a subcontractor will be remedied at the subcontractor's own expense. Neither the Owner's failure to note non-conforming Construction Work prior to Project Final Completion, nor its Final Payment under this Agreement, will waive any claim by Owner for correction of non-conforming Construction Work.
- 4.3.4 Legal and Insurance Services. The parties shall secure appropriate insurance given the joint interests of the parties, the waivers of liability within the Agreement and the desire to be fully protected from major errors and omissions. The PMT will consider and fairly apportion responsibility for all necessary insurance that covers matters not necessarily covered by the Project insurance.
- 4.3.5 PMT Participation and Deliverables. The Owner will actively participate with the PMT and PITS in further development of the Contract Task Matrix, a BIM execution plan, and other management protocols and tools, and will perform the Contract Tasks assigned to it in the Contract Task Matrix (Exhibit C), as amended.
- 4.3.6 Pre-Existing Hazardous Materials. Except to the extent that remediation of existing Hazardous Materials is included in the General Contractor's or a subcontractor's scope of work, Owner is responsible for all Hazardous Materials existing at the Project site and, except to the extent included within the risk registry, will separately contract for the remediation of pre-existing Hazardous Materials. For clarity, General Contractor will be responsible for any Hazardous Materials that it or its subcontractors bring onto the Project Site.

- 4.3.7 Other Obligations. In addition to the above, the Owner will provide the PMT with timely decisions necessary to support the Project Objective, make timely payments, and perform all other Owner-designated tasks as required by this Agreement.
- **4.4 Prime Consultant.** The Prime Consultant is the Architect of Record and is responsible for designing the Work except for those portions of the Work that are being designed through Design-Build Subcontractors. All Design Services provided by Prime Consultant and its subconsultants must be consistent with the Base Program and meet or exceed the standard of care in Section 4.4.1.
- 4.4.1 Standard of Care. The Prime Consultant and its subconsultants will perform the Design Services using that skill and care used by other competent licensed architects, engineers, and subconsultants skilled in designing projects in Ontario that are similar in size, scope, quality, and complexity. Design Services performed in connection with this Agreement must also be in accord with all Applicable Laws. If any of the Applicable Laws relevant to Design Services conflict, the Prime Consultant will consult with the PMT on how the conflict should be resolved. Nothing in this Agreement requires the Prime Consultant or its design subconsultants to perform any services outside its license or contrary to laws, codes, or regulations. The Prime Consultant and its design subconsultants are not responsible for the means, methods, sequences or procedures, or safety precautions, or programs in connection with the Construction Work.
- 4.4.2 Consultants. The Prime Consultant may retain design subconsultants to perform portions of the Design Services in accordance with Article 6. All design subconsultants will be appropriately licensed, will sign and seal all architectural or engineering documents prepared by them to the extent required by Governmental Authorities, will perform all services in accordance with the standard of care in Section 4.4.1, and in the case of engineering subconsultants, will be the Engineer of Record for their respective scopes of work. All subconsultants will be identified in the Project roster and incorporated into the Contract Task Matrix (Exhibit C). The Prime Consultant is responsible for managing its subconsultants and coordinating their respective services with services provided by Prime Consultant, other subconsultants, and any Design-Build Subcontractors.
- 4.4.3 Review and Integration of Design-Build Information. The PMT may elect to have Design Services performed under design-build subcontracts through the General Contractor. The Prime Consultant and its subconsultants will provide all applicable performance specifications and design criteria for design-build Work and will review the design documents prepared by the Design-Build Subcontractors for integration into the Prime Consultant's design and for conformance with the Prime Consultant's design intent and Base Program. All Design-Build Subcontractors will be identified in the Project roster and will also be incorporated into the Contract Task Matrix.
- 4.4.4 PMT Participation and Deliverables. The Prime Consultant will actively participate with the PMT and PITS in further development of the Contract Task Matrix, a BIM execution plan, and other management protocols and tools.
- 4.4.5 General Duties and Obligations. The Prime Consultant will perform all of its obligations under the Contract Task Matrix (Exhibit C), as amended, and the Agreement including, but not limited, to the following:

- (a) manage and coordinate all design submissions, questions, and responses to all applicable Governmental Authorities and all other reviewing and permitting agencies;
- (b) sign and affix its professional seal on all documents prepared by Prime Consultant and arrange for its professional subconsultants to do the same for all documents prepared by subconsultant[s], to the extent required by Governmental Authorities;
- (c) perform all Design Services required by the Project Objective and in accordance with Applicable Law and furnish all reports, affidavits, certificates, and other documents required by Governmental Authorities relating to those portions of the Project designed by the Prime Consultant and its subconsultants;
- (d) collaborate with the General Contractor to produce visualization aids such as renderings or computer generated video;
 - (e) utilize Target Value Design (Section 7.2.5);
- (f) actively collaborate with General Contractor, cross-functional teams, and other IPD Team Members throughout the design process to develop a complete set of coordinated Implementation Documents that meet or exceed the Base Program;
- (g) engage in "pull-based" planning for design production using the Last Planner ™ or equivalent system;
- (h) timely review and approve all Submittals for conformance with the Implementation Documents;
- (i) provide the PMT with timely decisions necessary to support the Project Objective, make timely payments, and perform all other Prime Consultant-designated tasks as required by this Agreement;
- (j) provide construction administration, problem solving and resolution of design issues during the Construction Phase in accordance with the Contract Task Matrix (Exhibit C), as amended; and
- (k) work with General Contractor to deliver a Record Model to Owner after Final Completion.
- **4.5 General Contractor.** The General Contractor is responsible for providing preconstruction services and performing the Construction Work in accordance with the Implementation Documents and for supervising, directing, managing the Construction Work in a manner that meets the Project Objective. The General Contractor will be considered to be the "constructor" for purposes of the *Occupational Health and Safety Act*, R.S.O. 1990, c. O.1, as amended. All Work provided by General Contractor must be consistent with the Project Objective and meet or exceed the standard of care in Section 4.5.1. General Contractor is responsible for implementing all collaboratively developed means, methods, sequences, and safety procedures related to the Construction Work.

- 4.5.1 Standard of Care. General Contractor will perform all Construction Work using its best skill and attention and in a timely, workman-like manner consistent with the degree of care and skill customarily exercised by prudent general contractors performing Construction Work on projects of similar size, scope, quality, and complexity in Ontario. To the extent that the General Contractor engages Design-Build Subcontractors, those subcontractors must comply with the standard of care in Section 4.4.1 for their respective Design Services except that Design-Build Subcontractors are also responsible for the means and methods, sequences, and procedures of their respective portion of the Construction Work. Nothing in this Agreement requires the General Contractor to perform any work or services outside its license or contrary to the laws, codes, or regulations of the Governmental Authorities.
- 4.5.2 Subcontractors. General Contractor will retain all subcontractors required for performance of its Work in accordance with Article 6, which includes Risk/Reward Subcontractors and Standard Subcontractors. Throughout this Agreement, references to subcontractors in lower case refer to all subcontractors, regardless of type. General Contractor is responsible for, and will review, supervise, coordinate, and manage, the subcontracted Work. All subcontracted Work must be performed in accordance with the standard of care in Section 4.5.1 and the Implementation Documents.
- (a) Design-Assist Subcontractors will actively engage in Target Value Design services (including set based design, alternatives, cost and constructability analyses) and will construct their respective portions of the Project.
- (b) Design-Build Subcontractors will provide Design Services for their respective scope of the Work through appropriately licensed design professionals who are responsible for the design documents prepared by the Design-Build Subcontractor. Design-Build Subcontractors will coordinate their Design Services with those of the Prime Consultant, its design subconsultants, and other Design-Build Subcontractors. All design-build documents will be stamped and signed by a registered professional engineer or architect who is licensed in Ontario in the appropriate discipline. The Design-Build Subcontractors will furnish all reports, affidavits, certificates, and other documents required by any Governmental Authority that are required by the applicable building codes, laws, and regulations governing those portions of the Project. The written subcontract for the Design-Build Subcontractor must expressly state that the Owner and other Parties are intended beneficiaries of the subcontract and are entitled to enforce all obligations and obtain all benefits under the subcontract, including any ownership rights. Design-Build Subcontractors will remain liable and responsible for all design-build Submittals.
- 4.5.3 PMT Participation and Deliverables. The General Contractor will actively participate with the PMT and PITS in further development of the Contract Task Matrix, a BIM execution plan, and other management protocols and tools.
- 4.5.4 General Duties and Obligations. The General Contractor will perform all of its obligations under the Contract Task Matrix (Exhibit C), as amended, and the Agreement including, but not limited, to the following:
- (a) assist the Prime Consultant during all phases of design by providing Target Value Design services (including set based design, alternatives, constructability and cost analyses) and by coordinating design information between Design-Build Subcontractors and the PITS;

- (b) actively collaborate with Prime Consultant, PITS, and other IPD Team Members throughout the design process to develop a complete set of coordinated Implementation Documents that meet or exceed the Base Program;
 - (c) support and collaborate in Target Value Design (Section 7.2.5);
- (d) manage Project information by using a Building Information Model or Models linked to Project cost and schedule databases and provide Model checking and validation, including clash detection and integration of subcontractor models and integration with Project schedules (4D);
- (e) collaborate with Prime Consultant to produce visualization aids such as renderings or computer generated video;
- (f) work with Prime Consultant to produce a Record Model to Owner after Final Completion of the Project;
- (g) work with the Prime Consultant to produce a Project manual that documents the management protocols, Co-location Plan, QA/QC plan, risk assessment/mitigation strategy, safety plan, work hours and work restrictions, communication protocols, specifications, Project terminology, forms and similar administrative and procedural requirements;
- (h) manage, perform, oversee, and direct all Construction Work in accordance with the Implementation Documents and Project Objective through Final Completion of the Project, including applying for Permits and maintaining the premises and surrounding area free from the accumulation of undue waste materials or rubbish and arranging for the clean up of the Project at the completion of Construction Work;
- (i) coordinate work with Owner's Separate Contractors, including integration of the Separate Contractors into project planning, scheduling, and coordination activities:
- (j) install, commission, and test as required, including ancillary construction necessary for installation, of all Owner furnished materials and equipment;
- (k) engage in "pull-based" scheduling for Construction Work using Lean project delivery methods to avoid advancing the construction beyond what is called for in the work plan or identified as workable back-log;
- (I) timely prepare and submit Submittals that are in conformance with the Implementation Documents;
- (m) provide the PMT with timely decisions necessary to support the Project Objective, make timely payments, and perform all other General Contractor-designated tasks as required by this Agreement;
- (n) propose a Quality Assurance/Quality Control ("QA/QC") plan for approval by the PMT, and implement the approved QA/QC plan;

- (o) document risks that could jeopardize the Project success in a risk registry (to be reviewed and updated by PMT) that identifies the risks in priority order and documents strategies to avoid, mitigate and monitor these risks;
- (p) maintain a daily log that summarizes the conditions and activities on the Project including, without limitation, a record of weather conditions, the number and types of workers on site, the subcontractors working, the areas where Construction Work is progressing and the Construction Work accomplished, problems encountered on the Project, significant visitors to the site, including inspectors and testing agencies, and other similar relevant information as the Owner may reasonably require;
- (q) construct the Project in accordance with the Implementation Documents, including installation of Owner-provided materials and equipment and, to the extent required, through use of Separate Contractors; and
- (r) commission the Project with the assistance of the Prime Consultant, other IPD Team Members, and the independent commissioning agent, if any.

5. PROJECT MANAGEMENT

- **5.1 Project Management Team.** The Project Management Team includes representatives of the Owner, Prime Consultant, and General Contractor. The PMT provides management-level guidance for collaborative planning, design, and construction of the Project to achieve the Project Objective. The PMT is responsible for all Project progress and for developing benchmarks, metrics, or standards for progress evaluation. The PMT will provide for appropriate representation of all of the interests of the ICL participants at the PMT level.
- 5.1.1 Authority and Responsibility. The PMT will manage and coordinate implementation of the Project Objective and provide direction to the Parties and to the PITS. The PMT is responsible for developing goals, strategies, protocols and tools for monitoring Project progress against schedule, budget and key performance indicators. These management tools will be used to track the status of the Project against achievement of the Project Objective. If it appears that the Project Objective is not likely to be achieved, the PMT will develop a recovery plan to enable the Project to achieve the Project Objective. Subject to the requirement that its decisions be unanimous, it is authorized to manage and direct the Project. The PMT is not, however, authorized to direct the actions of Owner's, Prime Consultant's, or General Contractor's employees and is not responsible for any failure of Owner, General Contractor or Prime Consultant to perform their respective obligations. PMT decisions are final and not subject to review or modification except by subsequent PMT action or Owner's Directive.
- 5.1.2 Reliable Participation. Fundamental to the success of the PMT is the willingness and ability of each member to participate reliably throughout the Project by providing dependable commitments, promises, and information in the best interest of the Project. The Parties commit to supporting the full engagement of their PMT representative and to providing the necessary resources to allow the PMT member to meet or exceed its commitments. The PMT will arrange for space for the Project Participants to physically work together on the Project and will develop a plan (eg. set up, stations, equipment, scheduling) for use of the co-location space.
- 5.1.3 Submittals and Shop Drawings. The PMT may establish any necessary procedures and protocols with respect to submission, review and archiving of Submittals and

Shop Drawings. Such procedures and protocols are may establish timing and processes, but are not to restrict the Prime Consultant in the substance of the review of the submission.

- 5.1.4 Interpretation of Base Program and Implementation Documents. The PMT has the sole authority to interpret the Base Program and Implementation Documents. All Reguests for Information ("RFI") or clarification will be processed in accordance with this Section 5.1.3. Questions, conflicts and issues regarding coordination and constructability should be resolved during design and before Implementation Documents are issues. RFIs should not be used as a substitute for collaboration. If the requesting and responding parties are able to resolve the issue through consultation, the responding party will produce the RFI documenting the solution before the date at which a delay will occur. If the requesting and responding parties cannot agree on a plan for resolving the issue before a delay will occur, the responding party must notify the PMT, who will promptly address the issue with the parties. All RFIs will be in writing to the General Contractor with a copy to the PMT and to the extent possible will be documented in BIM. If the RFI does not affect the Target Cost or the Milestones, the General Contractor will immediately post it on the web-based portal and notify all necessary parties for implementation in the field. Work related to a RFI or clarification that impacts the Target Cost or Contract Time may not proceed until it is discussed with the PMT and documented in a PMT Bulletin pursuant to Sections 5.1.7 and 5.3. The PMT will resolve questions, discrepancies, ambiguities, and other clarifications regarding the requirements of the Base Program and Implementation Documents according to this Agreement, the Project Objective, and in the best interests of the Project. A PMT Bulletin will be issued documenting the PMT's resolution of the issue.
- 5.1.5 Project Meetings. The PMT will hold Regular Meetings and Special Meetings as set forth below. The PMT will designate a person to schedule Regular Meetings, facilitate communications between the IPD Team Members, lead Regular and Special Meetings, prepare and distribute minutes of all PMT meetings, and prepare PMT Bulletins for execution by the PMT members.
- (a) Regular Meetings. The PMT will establish appropriate regular meetings as PMT and SMT deem best.. "Regular Meetings" will be held to review, discuss, and evaluate the current status of the Project with respect to design issues, cost, and schedule, and implement programs to improve overall Project performance. The Regular Meetings will be held separately from other meetings to assure proper management of the Project and encourage candor among the Parties. The Regular Meetings may include Senior Management Team members, as well as any other necessary IPD Team Members whose attendance is required by the PMT.
- (b) Special Meetings. "Special Meetings" may be requested by any PMT member to allow the PMT to address a matter of urgency. The Party requesting the Special Meeting will provide at least 1 Business Day's written notice, unless all PMT members agree to a shorter timeframe. Notice of a Special Meeting will identify the issues to be addressed. If a PMT member is not able to attend either a Regular Meeting or Special Meeting because of a scheduling conflict, an alternate member of the Party may be designated pursuant to Section 5.1.8.
- 5.1.6 Direct Communications. The PMT members and the Parties' employees are encouraged to communicate directly as necessary to efficiently manage the Project and to execute each individual PMT member's responsibilities. All decisions affecting design, cost, or

Contract Time must be made by the PMT jointly according to Section 5.1.7 and confirmed in writing in a PMT Bulletin.

- 5.1.7 Decision Making. PMT decisions must be by unanimous agreement. All PMT decisions that change the Target Cost, Contract Time or ICL will be documented through a Change Order per Section 12.4. If the PMT is unable to reach agreement, the PMT will refer the issue to the SMT under Section 5.2.
- 5.1.8 Parties' Representatives. The PMT will include representatives from each Party as identified in the Business Terms Sheet. Each Party will assure that its PMT representative attends all PMT meetings, has authority to act on behalf of the Party, and fulfills his or her responsibilities as a PMT member. The PMT may approve any member's designation of an alternate member, but any proposed replacement of a PMT member will be subject to the SMT's approval, which will not be unreasonably withheld.
- **5.2 Senior Management Team; Owner's Directive.** Each Party will be represented by SMT members who will act on its behalf with respect to the Dispute Resolution Procedures set forth in Section 16 and, upon request, to meet with the PMT at any Regular or Special Meetings. In addition, the SMT will periodically attend PMT Meetings to gain deeper familiarity with the Project and provide executive coaching and support to the PMT.
- 5.2.1 SMT Decisions. Any matters requiring SMT decisions or action will be decided by majority vote of the SMT, with Owner, Prime Consultant, and General Contractor each having only 1 vote. SMT Bulletins will be issued by SMT to reflect a decision by SMT. Despite the above, the Owner will have the right to override the majority decision of the SMT by issuing a written Owner's Directive. If an Owner's Directive causes the cost or duration of the Project to be increased, the Target Cost, Contract Time, and ICL, as applicable, will be adjusted pursuant to Article 12. Any dispute resulting from an SMT decision or an Owner's Directive will be resolved under the dispute resolution provisions set forth in Section 16. Each of the Parties' SMT representatives is set forth in the Business Terms Sheet. A Party may appoint a replacement SMT member by providing written notice to the other Parties.
- 5.3 Written Confirmation of Decisions. All decisions affecting design, cost, schedule, or reallocation of the Work will be recorded in writing by a PMT Bulletin per Section 5.1.6 and will be issued directly to the General Contractor or Prime Consultant for distribution to the appropriate subcontractors and subconsultants. PMT Bulletins must be signed by all PMT representatives to signify their concurrence with the decision. A Change Order that adjusts the Target Cost, ICL or Contract Time must be counter-signed by all PMT representatives and will be deemed a PMT Bulletin as well. All other decisions will be recorded through written meeting minutes that will be prepared by the person designated by the PMT pursuant to section 5.1.5. PMT decisions and meeting minutes will be circulated to the appropriate IPD Team Members and maintained on a collaborative web portal.
- **5.4 Personnel Management.** The PMT will not supervise or control any person the Owner, Prime Consultant, or General Contractor employs or contracts with in connection with the Project. The PMT may, however, require any Party to remove from the Project any person employed in connection with the Project, or personnel of companies that any Party contracted with, if the PMT determines that the presence of that person is detrimental to achievement of the Project Objective. The Owner, Prime Consultant, and General Contractor will provide personnel according to staffing plans approved by the PMT. The Parties will not

remove or reduce involvement of any personnel set forth in an approved staffing plan without the PMT's written consent, which will not be unreasonably withheld. Staffing plans may be prepared and approved in phases, as directed by the PMT.

6. SUBCONTRACTS AND CONSULTING AGREEMENTS

- **6.1 Written Agreements.** The Work performed under this Agreement will be executed by various Project team members, which will include subcontractors, suppliers, Vendors, and subconsultants. Consultants, subcontractors and Vendors may be Risk/Reward Team members or may be Standard Subcontractors or Standard Consultants. All subcontractors and subconsultants will be retained through subcontracts to the General Contractor or consulting agreements to the Prime Consultant. Risk/Reward Team members will be engaged through subcontracts, purchase orders (for Vendors), or consulting agreements that incorporate the risk/reward and liability allocation provisions of this Agreement. All Risk/Reward Team members retained by General Contractor or Prime Consultant will be engaged through agreements approved by the Parties, the terms of which may not be modified without the PMT's written approval.
- 6.1.1 Selection. Each Party will provide the PMT, in writing, with the names of persons or entities proposed to perform any portion of the Work. Within 5 Business Days of receipt, the PMT will provide a written response if it objects to the proposed person or entity because of the proposed person's or entity's qualifications, educational history, work history, or other reasonable basis. Failure of the PMT to object in a timely manner or recommend other potential subcontractors, suppliers, Vendors, or subconsultants will constitute the PMT's acceptance. Key subconsultants, subcontractors, Vendors, and suppliers will be procured early during the Validation Phase. Those subconsultants, subcontractors, Vendors, and suppliers that were not procured during the Validation Phase will be procured sometime during the Design/Preconstruction Phase or, in some cases, early during the Construction Phase.
- 6.1.2 Transparency in Negotiating or Bidding. Standard Subcontractors and suppliers will be selected on an open bid or negotiated basis as determined by the PMT. The process will be open and transparent with subcontractors and subconsultants submitting detailed breakdowns of their bids or proposals. Subcontractors, Vendors, and subconsultants that are Risk/Reward Team members will provide detailed information concerning their respective Chargeable Costs, profit included in the ICL, and the profit percentage cap for Change Orders. The ICL Distribution Spreadsheet (Exhibit G) will be amended to incorporate the Risk/Reward Team members' ICL and ICL Percentages. Standard Subcontractors and suppliers will provide detailed information about their respective bids or proposals including costs associated with the Work and overhead and profit for Change Orders.
- 6.1.3 Contract Award. Upon completion of the selection process for each subcontractor or subconsultant, the General Contractor or Prime Consultant will make a written recommendation to the PMT for contract award. If a PMT member objects to a person or entity on one of the bases set forth in Section 6.1.1, the objecting PMT member will propose another person or entity to which the PMT has no reasonable objection. A Party will not contract with a proposed person or entity reasonably and timely objected to by the PMT. Substitutions of a subcontractor or subconsultant previously selected and approved by the PMT will not be made without written PMT approval.

13

- **6.2 Licensing Requirements.** All subcontractors and subconsultants will be properly licensed for their respective portion of the Work.
- **6.3** Required Pass Through Provisions. The Parties must pass the following provisions through to subcontractors, suppliers, Vendors and subconsultants in their respective agreements as set forth below.

6.3.1 Risk/Reward Team Members

- (a) Waiver of Liability and Claims. The General Contractor and Prime Consultant will incorporate clauses substantially similar to Sections 13.1, 13.2, 13.3 and 13.4 regarding waiver of liability and claims against the Parties and other Risk/Reward Team members:
- (b) Indemnification. Substantially similar indemnity provisions to those set forth in Section 13.5 will be incorporated into Risk/Reward Team member subcontracts, Vendor agreements, and consulting agreements. The Owner, General Contractor, and Prime Consultant must be indemnified parties under all such provisions;
- (c) Compensation and ICL. Those subcontractors, Vendors, and subconsultants that are Risk/Reward Team members will be subject to substantially similar compensation and ICL provisions, adjusted for their respective proportionate ICL amounts, as those set forth in Articles 8 and 9;
- (d) Change Orders. Those subcontractors, Vendors, and subconsultants that are Risk/Reward Team members will be subject to similar provisions governing Contract Time under Article 11; and
 - (e) Those provisions required by Section 6.3.2.

6.3.2 All Subcontracts, Purchase Orders, and Consulting Agreements

- (a) Contract Flow-through. By appropriate written agreement, the General Contractor and Prime Consultant will require each subcontractor, supplier, and subconsultant, to the extent of the Work to be performed by the subcontractor, supplier, or subconsultant, to be bound to the General Contractor or Prime Consultant by terms of the Contract Documents, and to assume toward the General Contractor or Prime Consultant all the obligations and responsibilities that the General Contractor or Prime Consultant assumes toward the Owner through those documents. Each subcontract, purchase order, and consulting agreement will preserve and protect the rights of the Parties under this Agreement with respect to the Work to be performed by others so that subcontracting will not prejudice the Parties' rights. Where appropriate, the General Contractor and Prime Consultant will require each subcontractor, supplier, and subconsultant to enter into similar agreements with lower-tier subcontractors, suppliers, and subconsultants. In addition to this general flow-through requirement, General Contractor and Prime Consultant are specifically required to flow through the requirements in Section 6.3.2(b) through 6.3.2(g) as specified.
- (b) Assignment. All subcontracts, purchase orders, and consulting agreements will include assignment provisions allowing assignment by the Party to Owner after termination of this Agreement by Owner for cause pursuant to Section 17.3 and provided that Owner chooses to accept the assignment of the subcontract, purchase order, or consulting

agreement. If the subcontract, purchase order or consulting agreement is assigned to Owner as a result of a termination for cause, and the Work has been suspended for more than 30 days, the subcontractors', suppliers', or subconsultants' compensation will be equitably adjusted for increases in cost resulting from the suspension.

- (c) Insurance Insurance provisions equivalent to those applicable in Exhibit H will be included in the subcontracts and consulting agreements.
- (d) Indemnification. The General Contractor and Prime Consultant will each include indemnification provisions into the subcontracts and consulting agreements of each Standard Subcontractor and Standard Consultant that require the Standard Subcontractor and Standard Consultant to defend and indemnify the Owner, Prime Consultant, General Contractor, and all Risk/Reward Team members from all claims, damages, and liability arising out of or related to their respective work or services, including breach of contract, bodily injury, personal injury, and property damage, but only to the extent caused by the breach of contract or negligent acts or omissions of the Standard Subcontractor or Standard Consultant, or anyone directly or indirectly employed by any of them and for whose acts they may be liable. The SMT may waive or modify this requirement on a case-by-case basis.
- (e) Dispute Resolution Proceedings. General Contractor will require all subcontractors, Vendors, and suppliers, and Prime Consultant will require all subconsultants, to be subject to the dispute resolution proceedings in Article 16.
- (f) Records and Auditing. General Contractor will require all subcontractors, Vendors and suppliers, and Prime Consultant will require that all subconsultants be bound by the records retention and auditing provisions in Section 10.13.
- (g) Confidentiality and Non-Disclosure. General Contractor will require all subcontractors, Vendors and suppliers, and Prime Consultant will require all subconsultants, to be bound by the provisions of Section 18.2 prior to receiving any information regarding this Project.
- 6.4 Separate Contractors. The Owner may retain Separate Contractors to perform work or services related to the Work under this Agreement. The Owner is responsible for the timeliness and quality of the work and services of its Separate Contractors. The General Contractor and Prime Consultant will coordinate their Work with the work and services of the Separate Contractors to allow for smooth and efficient workflow and integrated work product. The Separate Contractors will coordinate their Work with the General Contractor and will conform to the General Contractor's Health and Safety policies. The General Contractor and subcontractors will cooperate with the Owner in resolution of any disputes or claims that may arise between the General Contractor and/or a subcontractor and Owner's Separate Contractor(s). All disputes will be resolved in accordance with Article 16 of this Agreement. To the extent that the General Contractor and/or subcontractor is unable to recover from Owner's Separate Contractor, Owner will be responsible.

7. RESPONSIBILITIES BY PHASE

7.1 Validation Phase. During the Validation Phase, the IPD Team, under the direction of the PMT, will prepare a Validation Study that defines the Base Program, Target Cost, and a Milestone Schedule for approval and acceptance by the Owner. The Target Cost,

as adjusted in accordance with this Agreement, is the design target during Target Value Design and is used to determine incentive adjustments to the ICL during the Design/Pre-Construction Phase. If the IPD Team cannot develop a design solution consistent with the Base Program and with a Target Cost that is at or below the Owner's Budget by the end of Validation, then this Agreement may be terminated pursuant to Section 17.1, except that the Risk/Reward Team Members will not be paid any profit for the work performed prior to the termination. Alternatively, the IPD Team may continue work through Target Value Design to attempt to establish a design solution with the Base Program and with a Target Cost that is at or below the Owner's Budget by October 1, 2016. If the IPD Team cannot develop such a solution by October 1, 2016, then the Owner may terminate this Agreement pursuant to Section 17.1, except that the Risk/Reward Team Members will not be paid any profit for the work performed prior to the termination.

- 7.1.1 Validation Study. The Validation Study together with Target Value Design will define the Base Program, Target Cost, and the Milestone Schedule. It will also:
 - include each item identified in the Validation Study;
 - include a detailed breakdown of all projected Chargeable Costs sufficient to allow PMT to confirm that there is a reasonable probability of delivering the Base Program for the Target Cost within the Contract Time.
- 7.1.2 Joint Site Investigation. As part of the Validation Study, the Owner and Risk/Reward Team members will conduct the Joint Site Investigation at or concerning the Project site to review all existing site information; perform investigations and surveys; document all site-related information needed to design and construct the Project; verify existing conditions within the Project site, including all points of connection; location of all utilities; and the accuracy of existing surveys and other documentation provided by the Owner. If a Party has participated in the Joint Site Investigation, it may rely upon the completeness and accuracy of the information provided by the Owner, to the extent that it is not contradicted by the Joint Site Investigation or any additional documents created or arising out of the Joint Site Investigation.
- 7.1.3 Base Program. The Owner and Risk/Reward Team members will jointly develop the Base Program, which will be part of the Project Objective and amended into the Agreement as Exhibit B-1. The Base Program should include large areas of needed work without detailing specific solutions. The Base Program will be based on the Owner's Project requirements and will further define the final quality, quantity, functionality, aesthetics, sustainability, and other Project requirements approved by the PMT. The Project scope will include all elements explicitly included in the Base Program and those that are reasonably inferable from the Base Program. The Added Value Incentive Items will not be included in the Base Program.

7.1.4 Risk Register...

(a) Risks that have been identified and priced pursuant to a risk analysis and included in the risk register, will be included in Target Cost. When a proposed method of dealing with the risk has been identified, the projected Chargeable Costs for the solution will replace the risk register for that risk in the Target Cost. If the projected Chargeable Cost for an item is less than what has been carried in the risk register, those remaining funds will be moved from the risk registry into the Cost Model and will be part of any surplus of funds available for shared savings. Once Construction begins, if a new risk is identified, it will be included in the risk

register, however the Target Cost will not be adjusted.

- 7.1.5 Project Milestones. The Owner and Risk/Reward Team members will jointly develop and document the Project Milestones based on the Contract Time set forth in the Business Terms Sheet. The Project Milestones will project Milestones for design and construction, including anticipated completion of design and procurement of permit(s), procurement of long lead items, projections for key Construction Work, Owner's occupancy requirements and projections for Substantial Performance and Final Completion. The Project Milestones are part of the Project Objective and will be amended into the Agreement as Exhibit B-4.
- 7.1.6 ICL Distribution Spreadsheet. The PMT will develop the ICL Distribution Spreadsheet and update it to include those Risk/Reward Team members procured during the Validation Phase per Article 6 and indicate the respective ICL and ICL Percentages of the Risk/Reward Team. The ICL Distribution Spreadsheet (and its updates) will be amended into the Agreement as Exhibit G.
- **7.2 Design/Preconstruction Phase.** During the Design/Preconstruction Phase, the PMT will oversee development of the Implementation Documents for the Project in accordance with the Base Program and will complete the design and preconstruction activities as set forth below.
- 7.2.1 Procurement of Other Project Team Members. During the Design/Preconstruction Phase, the remaining key subcontractors, Vendors, suppliers, subconsultants and Separate Contractors will be contracted by the Owner, General Contractor and Prime Consultant pursuant to Article 6. The Project roster will be updated by the PMT to include all IPD Team Members.
- (a) Update Contract Task Matrix. The PMT will update the Contract Task Matrix (Exhibit C) as required to include the necessary subcontractors, Vendors, suppliers, subconsultants and Separate Contractors procured during the Validation Phase that will be performing certain tasks. The Contract Task Matrix will be updated from time to time as the PMT deems necessary to reflect the status of the Project and assign necessary tasks to the most qualified IPD Team Member.
- (b) Early Release and Prefabrication. The PMT may authorize early release of certain systems, phases, or prefabrication of materials or equipment.
- 7.2.2 Project Implementation Teams. The PMT will develop, guide, and direct Project Implementation Teams that are responsible for designing and implementing the Project consistent with the Base Program, Target Cost, and Added Value Incentive Items. The PITS will be organized into interdisciplinary, cross-functional teams that will vary depending upon the stage of the Project. These cross-functional teams may be temporary for certain issues or may be formed for the duration of the Project. The cross-functional teams will collaborate regarding Project elements, including site use and improvements, selection of materials, building systems, and equipment. The PITS will meet regularly with the PMT throughout the Target Value Design process to evaluate functionality, constructability, sustainability, life cycle cost analysis, and Added Value Incentive Items. The PITS will not have decision making authority, but the PMT will actively seek and consider input and counsel of the PITS.

- (a) Cost Model. With the assistance of the PITS, the PMT will develop a detailed cost model based on the Target Cost breakdown and the profit of each Risk/Reward Team member. The cost model will track paid Chargeable Costs, incurred but unpaid Chargeable Costs, current estimate of future Chargeable Costs, the percentage of the Work completed, Value Added Incentive items, any ICL distributions, the predicted variance of each line item, and the projected variance from the Target Cost. The cost model will be updated by the General Contractor based on a schedule developed by the PMT and will be reviewed and approved by the PMT.
- (b) Project Scheduling. The PMT and PITS will jointly develop an initial pull-based schedule per Section 11.4 based on the design requirements and the Substantial Performance date set forth in the Milestone Schedule. The pull-based schedule will define the flow of developing the Implementation Documents, avoiding bottlenecks, or advancing aspects of the design beyond what has been anticipated and approved for any given time period. The PMT and PITS will jointly develop phase plans per Section 11.4.1 based on the initial pull-based schedule for submission and approval by the PMT. Under PMT direction, the applicable IPD Team Members will collaboratively create and update work plans for Design/Preconstruction Phase activities per Section 11.4.2. The Parties and applicable IPD Team Members will only perform Work that is approved on the applicable work plan or has been identified as workable backlog.
- 7.2.3 Building Information Modeling Execution Plan. Early in the Design/Preconstruction Phase , the PMT and relevant IPD Team Members will participate in a BIM workshop to establish a BIM execution plan that addresses the uses identified in the Base Program. The PMT will meet and develop detailed protocols and procedures for the use of BIM on this Project, which will be documented in the BIM execution plan that is incorporated into the Project Manual.. The PMT will select a Model Administrator, who is responsible for receiving modeling information from the Project Participants and incorporating the information into a master BIM. Each Party is responsible for maintaining any individual design or analysis models and providing their model information, at appropriate intervals, to the Model Administrator.
- 7.2.4 Building Information Modeling Ownership/Licensing. The Design Materials, including all BIM files, will, upon completion of the Project, become the property of the Owner and the Parties agree to provide the Owner, as a deliverable before Final Completion, the most recent version of all BIM files. Notwithstanding the above, the design elements that were created by the Parties, before execution of the Agreement, as extensions to commercially available building information modeling software remain the property of the respective Party that created the extension, regardless of whether it was used in a Model for this Project. The Parties grant a non-exclusive, irrevocable, perpetual license to each other to use any BIM information solely for the purpose of designing, analyzing and constructing this Project and for its future maintenance and remodeling. Provided the Party has not been terminated for convenience or cause, Owner grants each Party a non-exclusive, irrevocable, perpetual license solely for use or display of the Project BIM information for educational or promotional purposes.
- 7.2.5 Target Value Design Process. The IPD Team, under the direction of the PMT, will use the Target Value Design process to optimize and coordinate the design in accordance with the Base Program and endeavor to create additional value by identifying alternative systems, means, and methods to reduce capital expenditures and life-cycle costs, analyze and improve work-flow, improve constructability and functionality, provide more

operational flexibility, and endeavor to reduce the actual Chargeable Costs while maintaining or increasing the quality and overall function of the Project. To the greatest extent practicable, the General Contractor and the relevant IPD Team Members will provide continuous cost projections and rapid cost evaluation of proposed design alternatives to the PITS and PMT throughout the Target Value Design process to assist the IPD Team Members, and the PMT, in making informed decisions about proposed design solutions.

- 7.2.6 Innovation and Added Value Incentive Program. Owner wants to encourage innovation and to increase Project value without exceeding either (i) the Owner's Budget or (ii) the Target Cost. If savings are generated during the Design/Preconstruction Phase by reducing the estimated cost of the Base Program below the Owner's Budget and the Target Cost, the PMT will recommend including Added Value Incentive Items to the Base Program and the ICL will be adjusted in accordance with Section 9.8. The initial list of Added Value Incentive Items will be included by Amendment as Exhibit B-3. Owner may update the list of potential Added Value Incentive Items that it would favorably consider, and the PMT may recommend other items or innovations that Owner, in its sole discretion, may choose to have added to the Base Program as Added Value Incentive Items.
- 7.2.7 Government Regulations. The PMT in collaboration with the PITS will identify and determine the meaning and effect of all applicable building code provisions and other building restrictions and requirements of Governmental Authorities. The Prime Consultant, in conjunction with the General Contractor and any Design-Build Subcontractors, will assist the Owner with filing all required applications, drawings, specifications, calculations, or other documents required for permits.
- 7.2.8 Implementation Documents. The PMT will finalize design in accordance with Base Program and within Target Cost and obtain a permit for the Construction Work. The Implementation Documents list will be incorporated into the Project Objective as Exhibit B-5 through Amendment to the Agreement.
- 7.2.9 Target Cost and Milestones for ICL Distribution. The PMT will establish the Target Cost based on Implementation Documents and incorporate the Target Cost breakdown into the Project Objective as Exhibit B-2 through Amendment to the Agreement. The Target Cost cannot exceed the Owner's Budget and must include any Added Value Incentive Items elected by Owner and incorporated by the PMT during the Design/Preconstruction Phase. All ICL Milestone distributions will be measured against the Target Cost.
- (a) Update ICL Distribution Spreadsheet. The PMT will update the ICL Distribution Spreadsheet to include all Risk/Reward Team members' ICL and ICL Percentages, and will establish Milestone ICL distributions for the Construction Phase. The updated ICL Distribution Spreadsheet will be incorporated into the Agreement through Amendment as Exhibit G and will supersede all prior versions of Exhibit G.
- (b) Update Cost Model. The PMT will update the cost model to incorporate the Target Cost.
- (c) Update Milestone Schedule. The PMT will update the Milestone Schedule, if necessary, to incorporate all Construction Work reflected in the Implementation Documents.

- **7.3 Construction Phase.** During the Construction Phase, the remainder of the subconsultants, subcontractors, suppliers, and Separate Contractors will be procured and the Construction Work will be performed in accordance with the permitted Implementation Documents.
- 7.3.1 Building Information Model. The BIM will be continually updated throughout the Construction Phase in accordance with the BIM Agreement so that the Model is current with the actual "as-built" condition and when completed will be the Record Model.
- 7.3.2 Commissioning. The General Contractor, in conjunction with Prime Consultant and other necessary IPD Team Members, and Owner's maintenance personnel, will schedule and oversee the final testing and start-up of utilities, operational systems and equipment, and assist the Owner with the building commissioning. All inspections and testing will be conducted by the Owner and its third party commissioning agent, and required Governmental Authorities. During commissioning and before Final Completion, the General Contractor and other necessary participants will oversee the subcontractor operation, adjustment and balancing of all equipment, and training of Owner's employees in the correct operation and maintenance of equipment.
- 7.3.3 Close-Out. Before Final Completion of the Project, the General Contractor will provide Owner with all required as-built drawings, the Record Model, operation and maintenance manuals, references, warranties, attic stock, and keying schedules as required by the Contract Documents. If a document exists digitally, General Contractor will provide Owner with an electronic copy in the document's native format. If the document exists in hard copy, General Contractor will provide Owner with a hard copy in addition to a native format electronic copy.
- 7.3.4 Final Payment and Determination of ICL and Shared Savings. After Final Completion of the Project, the PMT will determine final ICL distribution and any shared savings for the Risk/Reward Team members per Section 9.9.2.
- 7.4 Substantial Completion. When the General Contractor believes that the Project or a Project Stage has completed Substantial Completion, it will notify the PMT who will review the entire Project or the Work performed in the Project Stage, as applicable. If the applicable Government Authority approves occupancy of the Project, or partial occupancy for the Project Stage per Section 7.4.1, and the PMT finds that Substantial Completion has been achieved, the PMT will issue a certificate of Substantial Completion. If the Project or Project Stage (as applicable) has not achieved Substantial Completion, the PMT will provide the General Contractor with a list of items that must be completed, repaired or replaced ("Punch List") before Substantial Completion and the General Contractor, after the General Contractor and/or the subcontractors have completed those items, will notify the PMT that the Project or the Project Stage is ready for re-inspection. Unless otherwise agreed, the certificate of Substantial Completion establishes the date when responsibility for security, maintenance, heat, utilities, damage to the Construction Work and insurance transfers from the General Contractor to Owner.
- 7.4.1 Partial Use and Occupancy. Upon 3 Business Days' written notice to the PMT, and provided that the Governmental Authorities having jurisdiction over the Project have deemed the Project Stage fit for legal use and occupancy, the Owner may occupy or otherwise make use of all or any part of that Project Stage's space or equipment before Substantial

Completion of the entire Project. The PMT will issue a certificate of beneficial occupancy describing those portions of the Project Stage deemed fit for legal occupancy by the Governmental Authorities and establishing the date for when the risk of damage to the Work in that area is transferred to the Owner.

7.5 Final Inspection and Acceptance.

- 7.5.1 Final Inspection and Owner Occupancy of a Project Stage. Before the Owner uses and occupies a Project Stage, the PMT will review all Work contained within the Project Stage and prepare a list of items that need to be completed, repaired or replaced ("Stage Punch List"). Correction of all Stage Punch List items to the PMT's satisfaction is a condition precedent to Final Completion of the Project Stage.
- 7.5.2 Final Completion and Owner Occupancy of the Project. After the General Contractor notifies the PMT that it believes the Project has achieved Final Completion, the PMT will review the entire Project and prepare a list of any items that require completion, repair or replacement ("Final Punch List"). Any Final Punch List items for previously completed and occupied Project Stages will be corrected under the warranty provisions of the Agreement. Correction of all Final Punch List items to the PMT's satisfaction is a condition precedent to Final Completion of the Project and Final Payment. When the PMT has determined that Project Final Completion has occurred, it will issue a certificate of Final Completion.

8. COMPENSATION

8.1 General. The Prime Consultant's and General Contractor's compensation for the Work is the sum of their respective Chargeable Costs incurred for the Project to the extent allowed in Exhibit E and Exhibit F (including overhead on the basis set forth in Section 8.1.1), and subject to Project success, their respective percentage of the adjusted and available ICL. Overhead is based on rates agreed with Owner and covers all costs incurred by Prime Consultant or General Contractor, of any nature, other than payable as Chargeable Costs, and may not include any amount of profit.

8.1.1 Overhead

Overhead for the Prime Consultant and Risk/Reward Subconsultants is part of their Chargeable Costs. The Prime Consultant's overhead is embedded in its agreed hourly rates (Exhibit E-1).

Overhead for the General Contractor and Risk/Reward Subcontractors is part of their Chargeable Costs. The General Contractor's overhead is paid as a fixed percentage on top of its other Chargeable Costs, at the fixed rate shown in the Business Terms Sheet. With PMT approval, certain overhead costs may be included in the agreed billable rates outside the overhead markup, provided that an expense category may not be charged in both a rate burden and the overhead mark-up.

- 8.1.2 Owner's Costs. The Owner's costs will be tracked separately by the Owner and are not included in the Target Cost. However, the following costs incurred by the Owner may be counted towards the Target Cost:
- (a) The Owner's costs resulting from damage to the Work itself prior to Substantial Performance caused by Prime Consultant, General Contractor, or anyone for

whom either is responsible, but only to the extent such costs are not paid by an insurer. This provision captures those Owner's costs resulting from damage to the Work other than the Chargeable Costs of Prime Consultant or General Contractor.

(b) The Owner's costs to third parties necessary for the completion of the Work, such as permitting agencies, inspectors, testing labs, or geotechnical subconsultants, to the extent those costs result from an error, omission, or delay caused by Prime Consultant, General Contractor, or anyone for whom either is responsible.

8.2 General Contractor Self-Performed Work

8.2.1 Competitively Proposed Trade Work

- (a) If Owner consents, General Contractor may propose to self-perform trade work on the same basis as approved subcontractors submitting competing proposals. General Contractor's proposal must specify, in addition to any other Owner required information, the proposed fixed profit and the proposed overhead percentage associated with the self-performed trade work. Owner will confer with the Prime Consultant, who will be responsible for reviewing the proposals, and the Owner will determine, in its sole discretion, whether to accept General Contractor's proposal to self-perform the trade work or the proposal of a subcontractor and may seek confirmation of the proposed overhead rate through independent audit
- (b) Owner reserves the right to reject all proposals and seek new proposals, or to reject all proposals and negotiate a contract for the trade work with General Contractor per Section 8.2.2 or its subcontractor, or with a Separate Contractor. If the trade work is awarded to General Contractor on a competitive, lump sum basis, General Contractor will separately invoice for the trade work on a percentage of completion basis as if the trade work was being performed by an independent subcontractor. However, General Contractor will not apply its overhead percentage (Section 8.1) to its self-performed, competitively bid trade work nor will General Contractor's ICL be increased by adding self-performed, competitively bid trade work.
- 8.2.2 Negotiated Trade Work. Owner, at its sole discretion, may choose to directly negotiate the performance of trade work with General Contractor for a fixed profit for that portion of the Construction Work. The fixed profit amount will be added to the General Contractor's ICL and the respective ICL percentages of the Risk/Reward Team members will be proportionately adjusted. This amount of overhead and profit awarded to General Contractor for the self-performed trade work is General Contractor's sole compensation for this work, which will not be marked up for any additional overhead or profit.
- **8.3** Incentive Compensation Layer. The ICL and the ICL Percentages of the Risk/Reward Team members will be set forth in Exhibit G, as amended. The ICL can be modified by Change Order and may be disbursed, if earned, during the Construction Phase based on the Milestone distributions also set forth in Exhibit G and subject to the provisions set forth in Article 9. Prime Consultant and General Contractor will each be responsible for paying their own respective Risk/Reward Team members their respective portion of any distributed ICL.
- 8.3.1 Scope Swaps. The PMT may choose to reduce the work scope of a IPD Team Member and correspondingly increase the scope of work of another IPD Team Member

to capture the benefits, without limitation, of innovative designs, processes, or technologies. If the PMT determines that the increased scope for a Risk/Reward Team member creates a significant financial inequity, it may use a portion of any savings from the innovative design, process, or technology to equitably increase the ICL of that Risk/Reward Team member, and then recalculate the ICL and the ICL Percentages of each Risk/Reward Team member. All changes under this section will be documented in a Change Order.

9. RISK/REWARD PROGRAM

- **9.1 General**. The Risk/Reward Team members have agreed that 100% of their potential profit (ICL) will be increased or decreased based on overall Project performance, as described in this Article. Although a portion of ICL may be paid at a Project Milestone, ICL is not earned until Final Completion of the Project and all adjustments to ICL have been made. No ICL increase for shared savings will be paid prior to Final Completion of the Project.
- **9.2 General Contractor's ICL Pass-through**. General Contractor's ICL is the sum of the ICL attributable to its Work and the ICL attributable to the Work of its Risk/Reward Subcontractors as set out in Exhibit G.

General Contractor will distribute to its Risk/Reward Subcontractors their respective proportions of the General Contractor's ICL. The ICL of subcontractors that join the Risk/Reward Team after execution of this Agreement will be added to General Contractor's ICL and all ICL Percentages adjusted proportionately.

- **9.3 Prime Consultant's ICL Pass-through**. Prime Consultant's ICL is the sum of the ICL attributable to its Work and the ICL attributable to the Work of its Risk/Reward Subconsultants as set out in Exhibit G. Prime Consultant will distribute to its Risk/Reward Subconsultants their respective proportions of the Prime Consultant's ICL. The ICL of subconsultants that join the Risk/Reward Team after execution of this Agreement will be added to Prime Consultant's ICL and all ICL Percentages will be adjusted proportionately.
- **9.4 ICL Distribution Schedule**. For each Project Stage, Exhibit G lists the Milestones and the percentage of a Risk/Reward Team member's ICL that is payable, subject to payment pre-conditions, at each Milestone. The ICL that may be distributed at a Milestone is based on the initial ICL amounts, and does not include any additional incentive compensation. The additional incentive compensation, if any, is distributed at Final Payment.

9.5 Milestone ICL Payments

- 9.5.1 When a Milestone is Achieved. When a Milestone is achieved, as preconditions to Milestone ICL payments, the PMT will assess whether (1) the Project is estimated to achieve Project Final Completion within the Contract Time and (2) the Estimated Final Cost is less than or equal to the Target Cost.
- (a) If both pre-conditions are true, the ICL Milestone payment scheduled in Exhibit G will be paid along with the Chargeable Costs for that period.
- (b) If either or both of the payment pre-conditions are not met, then the PMT will determine whether all or any portion of the scheduled Milestone payment of ICL will be deferred, and not paid. If at a later Milestone, both pre-conditions are met, the Milestone payment will include the scheduled Milestone payment plus the deferred Milestone payment or

payments.

- (c) Increases to the ICL due to Project performance are not included in Milestone ICL payments, but are only paid at Final Completion of the Project.
- Ontractor and Prime Consultant will prepare a proposed estimate that: (i) is current as of date of Milestone achievement; (ii) contains all elements and costs that are included within the Target Cost; and (iii) is based on the actual incurred Chargeable Costs plus the estimated cost to complete the Work. Within the next 5 Business Days after receipt of the proposed Milestone estimate, the PMT will either accept the proposed Milestone estimate or request verification from an independent cost subconsultant. If independent verification is requested, the PMT will meet with an independent cost subconsultant chosen by the Owner who will review the proposed Milestone estimate within the next 5 Business Days and either accept it, accept it with modification, or prepare an independent cost estimate. Any independent cost estimate prepared must follow the same structure as the proposed Milestone estimate and may not include any items not contained within Target Cost, increase any contingencies, or withhold any funds not previously agreed to by the PMT.
- 9.7 Milestone Overpayment. If, at any time during the Design/Preconstruction Phase or Construction Phase, the amount of the distributed Milestone ICL payments exceed the ICL, as adjusted, then Prime Consultant and General Contractor will reimburse the Owner their proportional share of any overpayment of ICL, including amounts of overpaid ICL that are paid to their subcontractors, Vendors, and subconsultants. Reimbursement will be made within 10 Business Days of receiving written demand from Owner. Also, if at any time after 50% completion of the Construction Phase the Owner reasonably determines that the aggregate Chargeable Costs are likely to exceed the Target Cost, the Owner may proportionally offset the applicable amounts of prior Milestone ICL payments against succeeding payments of Chargeable Costs to General Contractor and Prime Consultant respectively. Any such offset will be reversed after Project Final Completion if the actual Chargeable Costs do not exceed the Target Cost.

9.8 Design/Preconstruction Phase ICL Adjustments

- 9.8.1 General. Owner and the Risk/Reward Team intend to use Target Value Design processes to improve Project design, eliminate waste, and optimize Project value. The Design/Preconstruction Phase ICL adjustments are designed to promote innovation at the time when change is least disruptive to the Project and when savings can be most efficiently used by the Owner.
- 9.8.2 Added Value Incentive Items. Owner has developed a list of Added Value Incentive Items that it would like to include in the Project without increasing the Target Cost or Owner's Budget.
- 9.8.3 ICL Increase by Adding Value. If, through Target Value Design, the difference between the Estimated Final Cost and lower of (i) Owner's Budget and (ii) Target Cost is sufficient to cover [125]% of the estimated Chargeable Costs of an Added Value Incentive Item and Owner chooses to add the Added Value Incentive Item, then the ICL will be increased by [25]% of the estimated Chargeable Cost of the Added Value Incentive Item. Owner's election of any Added Value Incentive Item will be documented by Change Order

- 9.8.4 ICL Increase for Residual Design/Preconstruction Savings and Target Cost. When the PMT determines to establish the Target Cost, if the Estimated Final Cost (after including Target Value Design reductions and Added Value Incentive Item additions, if any) is less than the Owner's Budget, then the ICL will be increased by 50% of the difference. Target Cost will be set at the Estimated Final Cost.
- 9.8.5 Limit on Target Cost. The Target Cost cannot exceed the Owner's Budget , unless adjusted by Change Order.

9.9 Construction Phase ICL Adjustments

- 9.9.1 General. Owner and the Risk/Reward Team want to reduce rework, improve subcontractor productivity, and generally execute the Construction Phase as efficiently as possible. Thus, design innovation is not incentivized during this phase to avoid change that could disrupt construction efficiency.
- 9.9.2 Final Completion of the Project. If, at Final Completion of the Project, the Final Actual Cost (including reconciliation of allowances) is less than the Target Cost, then the ICL is increased by 50% of the difference. If after reconciliation of allowances the Final Actual Cost exceeds the Target Cost, then the ICL will be reduced, dollar for dollar, by the amount of the difference. If, after this reduction, the amount of ICL that had previously been distributed at Milestones exceeds the reduced ICL, then the Prime Consultant and General Contractor will, within 15 days of Project Final Completion, repay Owner the amount of the overpayment, in proportion to their ICL Percentages.

10. PAYMENT

- **10.1 Chargeable Costs.** The Prime Consultant's and General Contractor's Chargeable Costs are chargeable to the Target Cost. The Prime Consultant's Chargeable Costs are set forth in Exhibit E. The General Contractor's Chargeable Costs are set forth in Exhibit F. Consultants and subcontractors who are Risk/Reward Team members will use the same Chargeable Cost standards applied to Prime Consultant and General Contractor, respectively. No costs will be charged to the Project that are not actually incurred by the participants.
- 10.2 Periodic Payments. Payment applications will be prepared by the Prime Consultant and General Contractor in the format agreed by the PMT and submitted for approval no later than the 25th day of the month. The period covered by each payment application will be one calendar month. The payment application will include all Chargeable Costs incurred by the Prime Consultant and General Contractor for that calendar month. The PMT will approve all or part of the payment application within 5 Business Days of receipt. The Owner will make payment for all approved amounts within 30 Calendar Days of PMT approval. All payments will be subject to any applicable holdbacks required pursuant to the *Construction Lien Act*, R.S.O. 1990, c. C. 30, as amended, will only be released in accordance with the provisions of the Act.
- **10.3 Final Payment.** The Owner will make Final Payment 30 Calendar Days after Final Completion of the Project and determination of the ICL adjustments per Section 9.9.2. Final Payment will include all unpaid Chargeable Costs, plus the Risk/Reward Team members'

respective ICL amount, if any. The PMT must approve the payment application for Final Payment. Final Payment does not waive Owner's right to later object to defective design, materials or workmanship; waive any warranty rights the Owner may have; or release any Party from its indemnification obligations in Section 13.5. Payment will be subject to any applicable holdbacks required pursuant to the *Construction Lien Act*, R.S.O. 1990, c. C. 30, as amended. Holdbacks under the *Construction Lien Act*, R.S.O. 1990, c. C. 30, as amended, will only be released in accordance with the provisions of the Act.

- **10.4 Materials and Equipment.** Periodic payment applications may include materials and equipment delivered and suitably stored on-site for subsequent incorporation into the Construction Work or, with PMT's prior approval, suitably stored off-site at an insured warehouse that is approved by the PMT. Payment for stored materials and equipment will be conditioned upon compliance with Owner's procedures for protecting the Owner's interest and establishing Owner's title to the materials and equipment. General Contractor will not make advance payments to subcontractors or suppliers for stored materials or equipment without PMT approval.
- **10.5 Supporting Documents.** Each payment application will be accompanied by the following:
- 10.5.1 Sufficient documentation supporting the Chargeable Costs claimed in the payment application, including, without limitation, receipts, purchase orders, contracts, time reports, and other documentation reasonably required by the PMT or the Owner.
- 10.5.2 Duly executed conditional waiver and release forms complying with Applicable Law covering all Work performed during the billing period by the Prime Consultant, General Contractor, subcontractors, subconsultants, suppliers, or any other party entitled to record or serve a stop payment notice or *Construction Lien Act*, R.S.O. 1990, c. C 30, as amended.
- 10.5.3 Certification that Prime Consultant and General Contractor have no knowledge of any construction liens with respect to the Work performed by others and that all subcontractors, subconsultants, and suppliers have been paid to date or will be paid with the proceeds for Work covered under the payment application. If Prime Consultant or General Contractor has knowledge of any stop payment notices or construction liens, it will certify as to what it knows of the stop payment notices or liens and covenant to bond or otherwise discharge or expunge the stop payment notices or liens per Section 13.5.5.
- 10.5.4 In addition to the above, within 5 Business Days after receipt of Final Payment, the Prime Consultant and General Contractor will provide Owner a form duly complying with Applicable Law covering all services and work performed by them and by their subcontractors, subconsultants, suppliers, or any other party entitled to record or serve a stop payment notice under Ontario *Construction Lien Act*, R.S.O. 1990, c. C. 30, as amended, with respect to any labour, services, equipment, or material rendered or provided for the Project.
- **10.6 Right to Withhold.** Owner may refuse to approve a payment application or, because of subsequently discovered evidence or subsequent observations, may nullify the whole or any part of a prior payment application to the extent the PMT determines, or the SMT determines by majority vote, is necessary to protect Owner from loss arising out of or resulting from:

- 10.6.1 The Willful Default of the Party submitting the invoice;
- 10.6.2 Failure of General Contractor, or its subcontractors, to make timely payments to their respective subcontractors and suppliers for labour, materials, or equipment, provided Owner has made payment as required by this Agreement:
- 10.6.3 Failure of Prime Consultant, or its subconsultants, to make timely payments to their respective subconsultants for Design Services rendered in connection with the Project, provided Owner has made payment as required by this Agreement; and
- 10.6.4 Insufficient documentation, erroneous estimates of value of the Work performed, or other incorrect statements in a payment application.

If the reason for withholding payment is cured to the reasonable satisfaction of the PMT or the SMT by majority vote, Owner will pay the withheld amount within 20 days the PMT or SMT decision. The right to withhold does not apply to undisputed portions of any payment application.

- **10.7 No Right to Stop Work.** Subject to Article 17, if a Party disputes any determination with respect to any payment application, the Party will nevertheless expeditiously continue to prosecute the Work, if amounts approved by the PMT are timely paid. Owner will not be deemed to be in default or breach of this Agreement for withholding of any payment under Section 10.6. However, the Party claiming that payment has been withheld, in whole or in part, may submit the payment dispute for determination under Article 16.
- **10.8 Reliance.** In taking action on payment applications, the PMT may rely on the accuracy and completeness of the information furnished by the Prime Consultant and General Contractor and will not be deemed to represent that the PMT has made: (i) a detailed examination, audit, or arithmetic verification of the documentation or supporting data; (ii) exhaustive or continuous on- site inspections; or (iii) examinations to ascertain how or for what purposes the Prime Consultant and General Contractor have used amounts previously paid.
- 10.9 Warranty of Title. General Contractor warrants that title to all Construction Work, materials, and equipment covered by a payment application, whether incorporated in the Project or not, will pass to Owner at the time of payment by Owner, free and clear of all liens, claims, security interests or encumbrances in favor of General Contractor, subcontractors, suppliers, or other persons or entities entitled to make a claim by reason of having provided labour, materials, or equipment relating to the Construction Work. General Contractor will defend, indemnify, and hold Owner harmless from any and all liens, claims, security interests, or encumbrances filed by General Contractor, or its subcontractors, suppliers, or other persons or entities entitled to make a claim by reason of having provided labour, materials, and equipment relating to the Construction Work, provided General Contractor has received payment for the Construction work at issue pursuant to this Agreement. (See also, Section 13.3.5)
- **10.10 No Waiver.** Payment by Owner will not constitute approval or acceptance of any item of cost in the payment application or final acceptance or approval of that portion of the Work to which the partial payment relates.
- **10.11 Payments to Subcontractors and Suppliers.** The Owner and Prime Consultant do not have any obligation to pay, or to see that payment is made to a subcontractor or supplier except as may otherwise be required by law.

10.12 Payment to Consultants. The Owner and General Contractor do not have any obligation to pay, or to see that payment is made to a Prime Consultant's subconsultant except as may otherwise be required by law.

10.13 Records and Audits.

- 10.13.1 Prime Consultant and General Contractor must keep full and detailed accounts and records related to the performance of Work under this Agreement for a period of at least 7 years after Final Completion of the Project. The detailed accounts and records will be created and maintained in accordance with good accounting practices and control systems satisfactory to Owner. The records maintained must include, without limitation, job cost reports, general ledgers, time records, receipts, subcontracts, purchase orders, evidence of payment, written and electronic communications, estimates, bids, change order logs and files, change order request logs and files, and records related to any Chargeable Costs charged to the Project.
- 10.13.2 At any time during the performance of the Work and through 7 years after Project Final Completion, Owner, at its expense, may, but is not under any obligation to, audit the records of Prime Consultant or General Contractor, and their subconsultants and subcontractors of any tier, related to: (i) direct costs provided in establishing the Target Cost; (ii) any payment application or calculation of amounts the Owner owes or is alleged to owe; and (iii) subcontractor or subconsultant costs submitted as Chargeable Costs. The person or organization being audited will reasonably cooperate and make its records available for inspection, copying, and audit. If an audit discovers undisputed overcharges that the Prime Consultant or General Contractor knew or reasonably should have known and that exceeds 10% of any invoice, the Party responsible for the overcharge will reimburse Owner for the cost of that audit and if both the Prime Consultant and General Contractor are responsible for the overcharge, each will reimburse the Owner in proportion of the amount of the overcharge.
- 10.13.3 These audit provisions survive the termination of this Agreement. By appropriate written agreement, Prime Consultant and General Contractor will require their respective subcontractors, Vendors, suppliers, and subconsultants to be bound by and fully comply with the provisions of this Section 10.13.

11. CONTRACT TIME

- **11.1 Contract Time.** The Contract Time is the time allotted in the Milestone Schedule to achieve Project Final Completion. The Substantial Performance and Final Completion dates may only be extended by mutual agreement of the PMT in accordance with the Change Order process under Article 12.
- **11.2** Commencement of the Construction Work. The date of commencement of the Construction Phase will be documented by the Notice to Proceed issued by the PMT.
- 11.3 Project Staging. Portions of the Project may be segregated into separate Project Stages with individual Milestones, Substantial Performance, and Final Completion dates. If the Project is staged, all Project Stages must be completed prior to the date for Final Completion of the Project. The PMT will prepare a Staging Schedule for approval by the SMT that, at a minimum, specifies the dates of construction commencement, Substantial Performance, and Final Completion for each Project Stage.

- 11.4 Project Scheduling. All planning and scheduling performed on the Project will use "pull scheduling" techniques pursuant to the Last Planner System™, or an equivalent system. In order to be pull-based, the planning system must be based upon requests from a IPD Team Member to other IPD Team Members upon whom the requester's work is dependent, and receipt of reliable promises made by the upstream performer about when it will finish the work agreed upon per the hand-off criteria, in order to enable the downstream performers to begin their respective portion of the Work. At a minimum, the system must include the Milestone Schedule, collaboratively created phase plans per Section 11.4.1, make-ready work plans per Section 11.4.2, and a method for measuring, recording, and improving planning reliability.
- 11.4.1 Phase Planning. Phase plans must be based on collaborative planning through direct communications by all IPD Team Members performing Work in a particular phase, who, working backwards from the most current approved Milestone Schedule, create collaborative phase schedules indicating when their portion of the Work will be completed. Direct communications among IPD Team Members allows the various parties to make reliable promises to each other, and specifically discuss and negotiate the hand-off criteria or other conditions of satisfaction that are mutually understood and agreed upon.
- 11.4.2 Work Plans. As part of the phase planning, the PMT in collaboration with the PITS will establish a work plan to review upcoming design and construction performance requirements and establish the frequency of look ahead meetings and work plan schedules. The work plan schedules will document all Work performed during the prior week's period and project Work to be performed during the upcoming week(s). The work plans are to be used as a working tool to reflect commitments made in look ahead meetings, evaluate any upcoming constraints or schedule slippages, identify workable backlog and collaborate on methods for labour efficiency. Work flow will be scheduled to optimize the flow of Work through the Project and reduce bottlenecks and activities that will not advance the Contract Time or other Milestone dates.
- **11.5 Schedule Slippage.** The General Contractor or Prime Consultant will notify the PMT promptly of any delay in the most current approved Project schedule as a result of its Work and must submit a detailed recovery plan for the PMT to evaluate and approve.
- **11.6 Acceleration.** The PMT may determine that it is in the best interest of the Project to direct certain IPD Team Members to work overtime in an attempt to recapture any delays to the Project schedule. Overtime directed by a PMT Bulletin is a Chargeable Cost.

12. CHANGES

- **12.1 Permitted Changes.** There will be no changes to the Project Objective, including the Target Cost, Contract Time, or the ICL unless one of the following conditions occur:
 - 12.1.1 Owner-Elected Changes;
 - 12.1.2 Owner's Directive;
 - 12.1.3 Unforeseen and Differing Site Conditions;

- 12.1.4 Post Permit Changes;
- 12.1.5 Owner's suspension of the Work per Section 17.2; and/or
- 12.1.6 Reconciliation of allowance items to actual cost.

12.1.7 .

- **12.2 Changes to Cost.** Subject to PMT approval, the Target Cost may be adjusted by the estimated increase in Chargeable Costs caused by one of the conditions set forth in Sections 12.1.1 through 12.1.6 above.
- **12.3 Limitation on Changes to ICL.** A permitted change under Section 12.1 does not necessarily result in an adjustment to the ICL, which will be made only with the PMT's agreement and as set forth in a Change Order. However, if the ICL is adjusted, the ICL adjustment cannot exceed the product of (a) the ICL Change Order % applicable to a Risk/Reward Team member whose portion of the Work has increased and (b) the anticipated increase in Chargeable Costs of the Risk/Reward Team member.
- 12.4 Change Order Procedure. Change Orders will be used to document changes to the Project Objective and will include approved adjustment to the Target Cost, Contract Time, or ICL. Adjustments to the ICL must be tracked in accordance with the distribution chart set forth in Exhibit G for each approved Change Order. Any Party may request a Change Order to this Agreement for a permitted change by providing the PMT with a written Change Order Request setting forth the nature of the change, the reason for the change, and the effect, if any, on the Target Cost, the Contract Time, if any, or ICL. All CORs must be submitted to the PMT within 10 Business Days of discovering the condition or circumstance necessitating the change. Failure to timely submit the COR waives and releases any claim for a Change Order related to the facts or circumstances allegedly supporting the COR. The PMT will promptly review the COR and (i) accept the request, (ii) accept the request in part or with modification, (iii) request additional information or perform its own investigation, or (iv) deny the COR. If a COR is accepted by the PMT, then the PMT will recommend the Change Order to the Parties for execution. If the PMT does not act on a COR within 10 Business Days of its submission, it will be referred directly to the SMT. The SMT must act on the COR within 10 Business Days of referral from the PMT, and if it does not act, the Change Order is deemed denied by Owner's Directive. Any disagreements with regard to a COR will be determined according to the decision procedures in Sections 5.1.7 and 5.3, and if unresolved, are subject to dispute resolution under Article 16.
- **12.5 Owner's Directive.** When Owner issues an Owner's Directive, Prime Consultant and General Contractor must separately track their Chargeable Costs incurred in complying with the Owner's Directive and indicate on the applicable payment application(s) which Chargeable Costs are for a particular Owner's Directive.

13. LIABILITY ALLOCATION

13.1 Waiver of Liability. Notwithstanding anything to the contrary that may be contained in this Agreement, in no event shall any of the Risk/Reward Team members be individually liable to the Owner or any other entity affiliated with or related to the Owner for any damages, claims, demands, suits, causes of action, losses, costs, expenses and/or liabilities

regardless of whether such liability, guarantee or warranty, tort (including negligence), product liability, indemnity, contribution, strict liability or any other legal theory, in excess of such Party's individual ICL amount plus the amount of ICL received by such party except for the Allowed Claims listed in Section 13.2 and except as provided in Section 13.3.

13.1.1 The preceding Waiver of Liability shall not apply:

- (a) For the benefit or protection of any Risk/Reward Subcontractors, Risk/Reward Suppliers or Standard Subcontractors or suppliers to the extent that any acts, errors, omissions or breaches of contract by them would be covered under Subcontract Default Insurance procured for or covering subcontractors or suppliers on this Project.
- (b) To Subcontractor and supplier liability including Risk/Reward Subcontractor's and Risk/Reward Supplier's extending to claims paid under the Project's Subcontractor Default Insurance due to such Subcontractor and/or Supplier default or Risk/Reward Subcontractor or Risk/Reward Supplier default. In which case full subrogation right still apply against such defaulting Subcontractor and Suppliers.
- (c) To liabilities to the extent satisfied directly or indirectly from project-specific insurance proceeds actually received by or for the benefit of the Owner from Project Specific Insurance Policy, on the understanding that if any amount of such liabilities was paid for as a Chargeable Cost then such proceeds shall be added to the Target Cost and the amount allocated to and distributions from the Risk/Reward Team shall be re- calculated, regardless of whether the party directly paid by the insurer on the claim was the party that originally sustained the covered loss. In all cases, the deductible is considered a cost of work as outlined in Exhibit F.
- (d) For the benefit of a Risk/Reward Team member who has abandoned the Project, prior to having completed its scope of Work or Services, but only to the extent of liabilities arising directly from its abandonment of the Project.

13.2 Allowed Claims. "Allowed Claims" are limited to the following circumstances.

- 13.2.1 Willful Default. Claims for damages caused by the Willful Default of Prime Consultant, General Contractor, Vendor, subcontractors, or subconsultants;
- 13.2.2 Warranty Claims. Subject to Section.13.1, claims against the General Contractor, subcontractors, or suppliers for any breach of warranty obligations under Article 15;
- 13.2.3 Project Performance. Claims for loss or damage first occurring after Project Final Completion arising from (i) bodily injury or property damage caused by the failure of the Construction Work to be executed in conformance with the Implementation Documents; (ii) bodily injury or property damages caused by negligent acts, errors, or omissions in the design of the Project or its component systems; and (iii) the repair, modification, or replacement of components or systems that do not meet the functional and performance requirements of the Implementation Documents, provided that the liability for an Allowed Claim pursuant to this clause 13.2.3(iii) shall not exceed the Targeted ICL.
- 13.2.4 Third-Party Claims. Claims for contribution or indemnification from claims of third parties arising from any negligent acts or omissions of the Owner, Prime Consultant, General Contractor, subcontractors, suppliers, and subconsultants resulting from the performance

of the Work.

- 13.2.5 Unresolved Change Order Requests. Claims for Change Orders that have not been resolved by the PMT or SMT pursuant to Article 12.
- 13.2.6 Non-Payment. Claims arising from the Owner's failure to pay amounts that the PMT or SMT has approved for payment or amounts withheld by Owner pursuant to Section 10.6.
- 13.2.7 Over-Payment. Claims for amounts overpaid by Owner, as determined by audit under Section 10.13.
- 13.2.8 Termination or Suspension. Claims for amounts due following termination or suspension to the extent permitted by Article 17.
- 13.2.9 Indemnity. Claims to enforce indemnification obligations in Section 13.5, 13.6 or elsewhere in this Agreement;
- 13.2.10 Insurance. Claims that arise from the failure of the liable party to obtain insurance required by the Agreement.
- 13.2.11 Intellectual Property. Claims to enforce intellectual property rights under this Agreement.
- **13.3 Dispute Resolution/Specific Performance.** Nothing in Section 13.1 shall limit the rights of a party to enforce the dispute resolution provisions set forth in Article 16 and civil actions necessary to enforce construction liens of stop payment notice rights. Further Section 13.1 shall not limit a party from bringing an action for specific performance of this Agreement.
- 13.4 Limitation. If a claim is asserted against a party to this agreement who is not on the Risk/Reward Team or its insurer and that party or its insurer asserts a claim, cross-claim or third party claim alleging that the Risk/Reward Team member caused or contributed to the amount claimed from such party, the Owner agrees that the Owner will collect from such Risk/Reward Team member only those amounts that are (a) attributed to the fault, breach or liability of such Risk/Reward Team member and (b) not recoverable (by contribution, indemnity or other theory) by such party from the Risk/Reward Team member and that any costs incurred by the Risk/Reward Team member as a result of any such claim, cross-claim or third party claim by such IPD Team Member shall be a Chargeable Cost.

13.5 Indemnification.

13.5.1 General Contractor's Indemnification. The General Contractor will (and General Contractor will require its subcontractors and suppliers to) defend, indemnify, and hold the Owner and Risk/Reward Team members harmless from and against any and all claims, losses, damages, liabilities, and expenses (including legal, expert witness, and consulting fees and costs) alleged by any third parties (including any of their respective employees), arising out of, or resulting from, bodily injury (including death) or damage to tangible property (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the General Contractor, its subcontractors, and suppliers or anyone directly or indirectly employed by any of them for whose acts the General Contractor may be liable.

13.5.2 Owner's Indemnification. The Owner will defend, indemnify, and hold the Risk/Reward Team members harmless from and against any and all claims, losses, damages, liabilities, and expenses (including legal, expert witness, and consulting fees and costs) alleged by any third parties (including their respective employees) arising out of, or resulting from, bodily injury (including death) or damage to tangible property damage (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the Owner, its Separate Contractors, or anyone directly or indirectly employed by any of them for whose acts the Owner may be liable.

13.5.3 Prime Consultant's Indemnification. The Prime Consultant will (and Prime Consultant will cause its subconsultants to) defend, indemnify, and hold the Owner and Risk/Reward Team members harmless from and against any and all claims, losses, damages, liabilities, and expenses (including legal, expert witness, and consulting fees and costs) alleged by any third parties (including their respective employees) arising out of, or resulting from, bodily injury (including death) or damage to tangible property (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the Prime Consultant or its subconsultants, or anyone directly or indirectly employed by any of them for whose acts Prime Consultant may be liable.

13.5.4 Patent and Copyright. General Contractor and Prime Consultant each represent and warrant that designs, processes, methods, or materials used by each for the Project do not and will not violate any patents, copyrights, or trademarks. General Contractor and Prime Consultant, but only for its own breach of the prior sentence, will defend and indemnify Owner from and against claims, damages, losses, royalties and expenses, including, without limitation, attorneys' fees attributable to patent, copyright, or trademark violations.

13.5.5 Lien-Free Obligation, If any subcontractor, supplier, or subconsultant records or files, or maintains any action regarding a construction lien, stop payment notice, or lis pendens relating to the Work or the property where the Project is situated, the General Contractor or Prime Consultant, as applicable, will immediately vacate, discharge, extinguish, or expunge the construction lien, stop payment notice, or lis pendens, provided the Owner has paid the General Contractor or Prime Consultant for the Work and the General Contractor or Prime Consultant failed to pay their respective subconsultants, subcontractors or suppliers. If Prime Consultant or General Contractor fails to make timely payments to their respective subconsultants, subcontractors, and suppliers as required, the Owner may reasonably settle or bond over those claims, or take other actions necessary to prevent a default under any other agreement affecting the Owner's property or the Project, and General Contractor or Prime Consultant, as applicable, will upon written demand reimburse Owner for any substantiated amounts that were necessary to satisfy Prime Consultant's or General Contractor's obligation to satisfy or discharge any claim for lien or stop payment notice. The Prime Consultant and General Contractor will each indemnify and hold the Owner harmless from any claims filed by their respective subconsultants, subcontractors, or suppliers for enforcement of construction liens or stop payment notices, provided the Owner has made payment to the Prime Consultant and General Contractor for the Work that is subject to the claim. Nothing contained in this Section will be construed to require the General Contractor or Prime Consultant to vacate, discharge, extinguish, or expunge any valid construction lien, stop notice, lis pendens, or other claim arising from the Owner's non-payment or a valid dispute between the Parties or a dispute approved by PMT with a third party.

13.5.6 Party's Choice of Counsel. In all instances where there is an obligation to

defend a Party or to pay the cost of another Party's defense, the defended Party may reasonably reject counsel appointed for its defense and select counsel of its own choosing, which will be paid by the party obligated to defend that Party.

Site Safety. The General Contractor is solely responsible for training, initiating, maintaining, and supervising safety precautions and programs, including making all necessary filings and providing all necessary notices under Applicable Laws, in connection with performance of the Construction Work and according to all Applicable Laws governing safety on construction sites. The General Contractor will maintain on-site all Material Safety Data Sheets ("MSDS") for material or products brought to the Project Site that are regulated under Occupational Health and Safety Act, R.S.O. 1990, c. O. 1, as amended, or any other Applicable Law regulating material or products. The General Contractor will defend, indemnify, and hold the Owner and Prime Consultant harmless from and against all demands, causes of action, and other claims for damage, loss, and expense, including but not limited to attorneys' fees, resulting from bodily injury, sickness, disease, death, injury, or tangible property damage (other than to the Work itself) to the extent caused by its failure to train, initiate, maintain, or supervise safety precautions and programs in connection with performance of the Construction Work. Except for fines or penalties resulting from violations arising from Owner's Separate Contractors' work, the General Contractor and each subcontractor is responsible for the payment of all fines levied against it or against Owner arising from or related to violation of Applicable Laws. These fines and penalties are not Chargeable Costs unless approved by the PMT.

Unless specifically required by the Contract Documents, or reasonably required by the construction process, the General Contractor and subcontractors may not use or introduce to the Project site, or incorporate into the Project, any Hazardous Materials. If Hazardous Materials are specifically required by the Contract Documents, the General Contractor or subcontractor will take all necessary precautions and comply with all Applicable Laws to protect workers and the public from deleterious exposure to the Hazardous Materials and will properly and lawfully dispose of any residual Hazardous Materials. If Hazardous Materials are released at or from the Project site, the observing General Contractor or subcontractor will immediately notify the PMT and the General Contractor will timely notify all Governmental Authorities having jurisdiction over the release.

Prior to the commencement of Construction Work, the General Contractor shall submit to the Owner:

- 13.6.1 a current Workplace Safety & Insurance Board Clearance Certificate;
- 13.6.2 copies of the General Contractor's insurance policies having application to the Project or certificates of insurance, at the option of the Owner;
- 13.6.3 documentation setting out the General Contractor's in-house safety programs; and
- 13.6.4 a copy of the Notice of Project filed with the Ontario Ministry of Labour naming itself as "constructor" under the *Occupational Health and Safety Act*, R.S.O. 1990, c. O. 1, as amended.

The Owner and Prime Consultant and through them all subconsultants agree that their personnel will comply with the policies and procedures of and the directions and instructions

from the General Contractor with respect to occupational health and safety and related matters. Prior to admission to the site of the Project, the General Contractor may, as a condition of admission, require any subcontractor or subconsultant to sign a written acknowledgement in the following form:

Acknowledgement

The undersigned acknowledges that the Work it will perform on behalf of the Owner requires it to enter a place of work which is under the total control of a General Contractor that has a contract with the Owner, pursuant to which the General Contractor has assumed overall responsibility for compliance with all aspects of the applicable health and safety legislation, including all the responsibilities of the "constructor" under the *Occupational Health and Safety Act*, R.S.O. 1990, c. O.1, as amended, as well as responsibility to co-ordinate and schedule the activities of our Work with the Work of the General Contractor under its contract. The undersigned agrees to comply with the General Contractor's directions and instructions with respect to health, safety, co-ordination, and scheduling and acknowledges that its failure to do so will be cause for termination of employment or of the undersigned's contract with the owner, as the case may be. The undersigned also agrees to have the General Contractor named as an additional insured on any comprehensive liability insurance policy, where such insurance is required.

13.7 Joint Defense of Third-Party Claims. Because the Parties have a similar interest in the outcome of the Project, the Parties will endeavor to resolve any third-party claims (including subcontractor, supplier, and subconsultant claims) under a joint defense agreement establishing the procedures and rights of the Parties. To the greatest extent possible, the Parties will jointly address, investigate, manage, defend, settle, or otherwise resolve all third-party claims arising from or related to the Project or this Agreement, subject to applicable legal and ethical considerations, including the need for independent legal counsel. Subject to PMT approval, a Party defending a claim on behalf of the Project may submit its reasonable expenses of defense as a Chargeable Cost.

14. PROJECT INSURANCE

- **14.1 General.** The Parties will carry the insurance required of them in Exhibit H and require their respective subcontractors, subconsultants, and suppliers to carry the insurance coverages indicated in Exhibit H. The availability of insurance does not limit the indemnification responsibility of any Party or Risk/Reward Team member, nor limit any other remedy available to a Party Risk/Reward Team member.
- 14.2 Bonding or Subcontractor Default Insurance. To the extent required by the PMT, the General Contractor will cause certain subcontractors to be covered under subcontractor default insurance, or to furnish payment and performance bonds in full compliance with Applicable Law for the full subcontract amount, with the additional costs adjusting the Target Cost through executed Change Order. To the extent required, the actual cost of the required bonds or subcontractor default insurance will be paid by Owner as a Chargeable Cost.

15. WARRANTY

15.1 Warranty and Correction of Work. General Contractor warrants that all Construction Work will be of good quality, free from defects, and conforming to the Implementation Documents and Applicable Law. For a period of 1 year commencing from

Substantial Performance of the Project or the date of repair of the defective component, whichever is later, and for longer periods specified in the Implementation Documents for certain equipment manufacturers or suppliers, General Contractor will repair or replace at its expense any and all deficient or defective Construction Work together with any other work that is damaged during repair or replacement. If a portion of the Construction Work is completed and accepted by Owner, then the warranty will commence from the date of acceptance or the date of repair of the defective component, whichever is later. General Contractor's warranty excludes improper or insufficient maintenance, improper operation, and normal wear and tear under normal usage. General Contractor will procure all subcontractor and manufacturer express warranties required under the Implementation Documents on the Owner's behalf and will transmit the warranties to Owner before Final Completion and Project close-out. Establishment of the 1 year express warranty period for correction of Construction Work relates only to the General Contractor's specific obligation to correct defective or non-conforming Construction Work, and is subject to the limits of liability set forth in Section 13.

16. DISPUTE RESOLUTION

- **16.1 Scope.** All disputes between the Parties and their respective Separate Contractors, subcontractors, suppliers, and subconsultants arising from or in connection with this Agreement will be resolved as provided in this Section.
- **16.2 Continued Performance.** Subject to the conditions of Article 17, at all times during the pendency of a dispute or a dispute resolution proceeding, Work will continue. If the Owner continues to comply with its obligations under this Agreement, the parties to the dispute resolution proceeding will continue to comply with any Owner's Directives.
- **16.3 Permitted Disputes.** No Party may bring any action, suit, or other proceeding against any other Party to this Agreement relating to or arising from this Agreement, except for (i) Allowed Claims, (ii) proceedings to seek an interpretation of the terms of this Agreement, (iii) proceedings to determine the Contract Time, or (iv) proceedings to determine Change Order disputes, ICL, or shared savings distributions.
- **16.4 Notice.** A Party may initiate the dispute resolution procedures stated in this Section by providing all PMT members with written notice of a potential dispute which specifies, in reasonable detail, the basis of the dispute and the remedy sought. Notice of a dispute must be given within 10 Business Days following the occurrence of the event or condition or circumstance giving rise to the dispute or the claim is waived. Under no circumstances will a claim be made if it is barred by applicable statutes of limitation.
- 16.5 Senior Management Team Meeting. If the PMT is unable to resolve the dispute, any party to the dispute may request the SMT members to meet with the PMT and attempt in good faith to resolve the dispute. The SMT will then review the claim in detail and meet face-to-face to discuss and resolve the matter in a "SMT Meeting." The SMT Meeting will occur no later than 10 Business Days after the PMT has declared an impasse in its efforts to resolve the dispute, unless the parties to the dispute agree upon a longer period of time. The SMT Meeting will be for the express purposes of: (i) exchanging and reviewing all pertinent non-privileged documents and information relating to the matters and issues in dispute; (ii) freely and candidly discussing each party's position; and (iii) reaching agreement upon a reasonable, compromise resolution of the dispute. If a negotiated settlement is reached, the terms of the settlement will be recorded in a written Amendment or Change Order signed by the parties to

the dispute.

- 16.6 **Mediation.** If the dispute is not resolved by the conclusion of the SMT Meeting, any party to the dispute may request mediation of the dispute in writing. If the Parties agree to mediate, they will jointly select a mediator acceptable to all SMT members and who is a construction attorney with at least 15 years of experience in resolving disputes for complex construction projects. Each Party to the dispute will give the mediator a written statement regarding the asserted claims, and the mediator may inspect the Project site, Project Objective, Implementation Documents, and other information reasonably required to understand the factual and legal basis of the dispute. The mediator will schedule a mediation session within 20 Business Days of being retained by the Parties. Representatives from each party who have authority to resolve the dispute, together with any other party who has an interest in the dispute, will attend the mediation. The parties involved in the dispute will bear the cost of mediation equally. The cost of mediation will not be a Chargeable Cost. The mediation proceeding will be confidential and not admissible in court. The mediation process must be completed within 40 Business Days of the mediator's being retained, unless all parties involved in the dispute extend the mediation period. If, as a result of the mediation, a negotiated settlement is reached, the parties will enter into a written settlement agreement that will be enforceable in a court of competent jurisdiction.
- **16.7 Litigation.** All disputes that are not either resolved through the informal process under Section 16.5 or by mediation under Section 16.6 after having undergone those processes will be subject to arbitration in accordance with the *Arbitration Act, 1991,* S.O. 1991, c. 17, as amended. \

17. DEFAULT, SUSPENSION AND TERMINATION

- 17.1 Termination for Convenience. The Owner may terminate this Agreement for convenience upon 10 Business Days prior written notice at any time. Subject to receiving the payment required by Section 17.1.2, the Prime Consultant and General Contractor each grant title to Owner for all Design Materials prepared prior to a termination under this Section 17.1 for Owner's use, at its discretion, for future continuation of the Project and for its future maintenance and remodeling. If the Project is terminated for convenience before commencement of the Construction Phase and the Owner uses the Design Materials for continuation of the Project following the termination, Owner will release, indemnify, defend, and hold harmless General Contractor and Prime Consultant from any liability arising from Owner's further use of these materials.
- 17.1.1 Notice. The notice will state the extent and effective date of the termination, and, on the effective date, the terminated Party will: (i) to the extent directed, stop Work under this Agreement; (ii) terminate or assign all subcontracts, purchase orders, and consulting agreements to Owner unless otherwise directed; and (iii) take other actions as may be necessary or reasonably requested by Owner to protect and preserve the Work and any other property in the terminated Party's possession in which Owner has or may acquire an interest.
- 17.1.2 Payment Upon Termination for Convenience. If the Owner terminates this Agreement for convenience, the Owner will pay the Prime Consultant and General Contractor: (i) all Chargeable Costs incurred prior to the effective date of termination; (ii) reasonable expenses related to the termination, including demobilization, unavoidable

subconsultant and subcontractor costs, and securing the Project site; and (iii) provided the Termination for Convenience is not pursuant to Section 7.1 whereby no ICL shall be payable, the lesser of a portion of the unadjusted ICL per Section 8.3 based on the percentage of the Work completed prior to termination or the amount of ICL payable at the next nearest Milestone ICL distribution as determined by an independent assessor prior to termination. In no event will the total amount paid to the Prime Consultant and General Contractor, inclusive of ICL, exceed the Owner's Budget or Target Cost, as applicable. Any payment under this Section is subject to Owner's receipt of all documentation required for payment under Section 10.5, subject to withholding by Owner for reasons and in the manner provided in connection with Final Payment. Any dispute over the amount to be paid upon termination will be resolved under the dispute resolution procedures in Article 16.

17.2 Suspension. The Owner may, without cause, order the PMT to suspend, delay, or interrupt the Project for as long as the Owner may determine. If the Project is suspended pursuant to this Section, the Contract Time will be extended. If the Project is suspended after issuance of the Notice to Proceed with the Work and the suspension results in an increase in the Chargeable Costs for the Work, either the Target Cost (depending on the phase) will be increased by the increase in Chargeable Costs reasonably caused by the suspension, and each Party's ICL, which will be increased by the product of its Change Order % multiplied by the increase in its Chargeable Costs due to the suspension. No adjustment will be made to the extent that performance was suspended, delayed, or interrupted by acts or omissions of the Prime Consultant, General Contractor, or any entity or persons working directly for either for which they are responsible. A suspension lasting 6 months will be deemed to be a Termination for Convenience.

17.3 Owner Termination for Cause.

- 17.3.1 Notice. The Owner may terminate a Party to this Agreement, upon 10 Business Days prior written notice, and unless otherwise agreed, an additional 10 Business Days to cure, if any of the following conditions occur. The effective date of termination will be 20 Business Days from the date of the notice. For clarity, a Party that is not in default of this Section cannot be terminated hereunder because of the default of another Party of this Section.
- (a) Failure of one or more Parties to this Agreement to provide adequate labour and resources to achieve the Project Objective;
- (b) Refusal by a Party to rectify Work that is not in accordance with this Agreement, the Project Objective, or Implementation Documents;
- (c) Failure of a Party to collaborate with the PMT for the benefit of the Project;
- (d) Failure of the Prime Consultant or General Contractor to properly pay its subcontractors, suppliers, and subconsultants;
 - (e) Bankruptcy or insolvency of a Party to this Agreement; or
 - (f) Acts of Willful Default by Prime Consultant or General Contractor.
- 17.3.2 Ownership of Design Materials in Termination for Cause. The Prime Consultant and General Contractor each grant title to Owner for all Design Materials prepared prior to a termination under this Section 17.3 for Owner's use, at its discretion, for future continuation of the Project and for its future maintenance and remodeling.
- 17.3.3 Owner Remedies. If Owner terminates a Party for cause, the terminated Party will not be entitled to any further payments until after Final Completion of the Project. The terminated Party will forfeit any ICL not yet paid to it and shall be responsible for any damages resulting from the default leading to the Termination for Cause up to a maximum of 100% of the Target Cost. In addition, Owner may, without prejudice to any other rights or remedies, and after giving the Party and its surety (if any) prior written notice:

- (a) Take possession of the site and of all materials and equipment procured for the Project;
- (b) Accept assignment of any purchase orders, subcontracts, and consulting agreements, as well as rental agreements for construction equipment and machinery at the site; and
- (c) Finish the Work by whatever reasonable method Owner may deem expedient, subject to its obligation to mitigate its damages.
- 17.3.4 Conversion to Termination for Convenience. If a court of competent jurisdiction deems that termination of the Party was wrongful or otherwise improper, the termination will be deemed a termination for convenience under Section 17.1. Owner will pay the terminated Party the amount due as if the termination had been for convenience, plus interest at 3 percent simple interest per annum, on the difference between the amount that was paid at termination and the amount that should have been paid for a termination for convenience, if any, from the date the payment would have been due to the date payment is actually made.
- 17.4 Prime Consultant/General Contractor Termination for Cause. The Prime Consultant or General Contractor may terminate this Agreement for cause upon 10 Business Days prior written notice, and unless otherwise agreed, an additional 10 Business Days to cure if any of the following occur: (i) Owner fails to pay amounts approved for payment by the PMT or SMT; (ii) after commencement of construction, Owner suspends the Project under Section 17.2 for more than 90 consecutive Calendar Days; or (iii) Owner has repeatedly failed to fulfill its obligations under the Agreement constituting an abandonment of the Project, or bankruptcy or insolvency of Owner.
- 17.4.1 Effective Date. Termination is effective 20 Business Days from date of notification (or such longer time as is reasonable provided the party in breach of Section 17.3 is remedying the applicable default in good faith).

17.4.2 Payment. If the Agreement is terminated under Section 17.4, the Owner will pay the Prime Consultant and General Contractor the: (i) Chargeable Costs incurred by the Prime Consultant and General Contractor prior to the effective date of termination; (ii) reasonable expenses related to the termination, including demobilization and securing the Project site; and (iii) if the termination is occurring after the Construction Phase has commenced, the lesser of a portion of the unadjusted ICL per Section 8.3 based on the percentage of the Work completed prior to termination or the amount of ICL payable at the next nearest Milestone ICL distribution as determined by an independent assessor prior to termination. The total amount paid to the Prime Consultant and General Contractor may not exceed either the Target Cost (depending on the phase). Any payment under this Section is subject to Owner's receipt of all requested statutory lien waiver and release forms, as well as other documentation required for payment under Section 10.5, subject to withholding by Owner for reasons and in the manner provided in connection with Final Payment. Any dispute over the amount to be paid upon termination will be resolved under the dispute resolution procedures in Article 16.

18. MISCELLANEOUS PROVISIONS

18.1 License. The Prime Consultant and General Contractor represent that they are properly licensed as required by the Province of Ontario to perform the Work required under this Agreement, and that each Party's business entity is in good standing and qualified to do business in the Province of Ontario.

18.2 Confidentiality

- 18.2.1 Parties' Financial Information. The Owner and Risk/Reward Team members will be disclosing information concerning their methods of accounting, pricing of products and services, and other confidential information. Each Party agrees that it will keep such information confidential and will only provide it to other persons or entities as reasonably necessary for the design or construction of the Project. If a Party makes a written demand for return of confidential information, the other Parties will make reasonable attempts to return or destroy any copies of the confidential information in their possession. Each Party will place similar confidentiality restrictions in any agreements with its Risk/Reward Team members of any tier.
- 18.2.2 Owner's Proprietary Information. Information provided by Owner or made available to the Parties during performance of this Agreement will not be disclosed to persons or entities other than as necessary to perform the Work or Design Services required by this Agreement.
- 18.3 Compliance With Anti-Corruption Laws. Prime Consultant and General Contractor (including their respective subsidiaries and affiliates and any directors, officers and employees of each such entity) each represent and warrant that they have each complied, and will continue to comply, with all applicable anti-corruption laws in connection with their respective Work under this Agreement.
- **18.4 Notices.** Any notice required to be given by this Agreement will be in writing and deemed effective upon: (a) the date of personal delivery if received by the addressee before 5:00 p.m. local time on a Business Day (if received after 5 pm it will be deemed to have arrived on the next Business Day); (b) 3 Business Days after being sent via registered or certified mail

with a return receipt requested; or (c) 1 Business Day after being sent by overnight commercial courier providing next-business-day delivery. Notices will be addressed to the following respective Parties:

Owner:	Prime Consultant:	General Contractor:

- **18.5 Governing Law.** This Agreement will be governed and construed under the laws of the Province of Ontario without giving effect to any choice of law or rule of conflict that would cause the application of the laws of any other jurisdiction. Venue for any action will be in Province of Ontario.
- **18.6 Commencement of Statute of Limitations.** Causes of action between the Parties to this Agreement pertaining to acts or failures to act will be deemed to have accrued and the applicable statutes of limitations will commence to run not later than either the date of Project Final Completion or, if applicable, the date of a recorded notice of completion, whichever is later.

- **18.7 Assignment.** The Parties respectively bind themselves, and their respective successors, assigns, and legal representatives to the other Parties to this Agreement. The General Contractor and Prime Consultant may not assign this Agreement without the express written consent of the Owner, and any unconsented assignment will be void.
- 18.8 Notice of Dissolution and Other Change of Business. If at any time during the performance of the Work and for a period of two (2) years following the completion of the Work: (i) Dissolution of any Party occurs, or (ii) any Party experiences or anticipates a material change in business, including a reorganization, refinancing, restructuring or leveraged buyout, or (iii) any Party has notice of any of the matters referred to in (i) or (ii); then that Party must furnish the other Parties with written notice of the same within 5 days and, in the case of Prime Consultant or General Contractor, take all necessary and proper safeguards to allow Owner the opportunity to retrieve its property, the Work and anything relating to the Work from their then- current location. For the purposes of this provision, "Dissolution" means, in respect of any Party, the making of an assignment for the benefit of creditors or admitting in writing its inability to pay its debts as they mature, bankruptcy, insolvency, liquidation, winding-up, administration and dissolution and the appointment in respect of it or any of its assets of a receiver, administrator, manager, or similar officer, and any proceeding or event which is equivalent or analogous to any of the same by whatever name known and in whatever jurisdiction and any step taken for or with a view to any of the foregoing.
- **18.9 Severability.** The terms and conditions of this Agreement will be interpreted according to their plain meaning, and not strictly for or against any Party. Any rule of construction or interpretation to the contrary will be of no force or effect with respect to this Agreement. If a court of competent jurisdiction finds any term or provision of this Agreement to be void or unenforceable for any reason, the term or provision will be deemed severed, and the remainder of the Agreement will remain in full force and effect according to its terms and provisions, to the maximum extent permitted by law.
- **18.10 No Third-Party Beneficiaries.** Nothing contained in this Agreement creates a contractual relationship with, or a cause of action in favor of a third party against any of the Owner, Prime Consultant, or General Contractor. The Parties acknowledge and agree that the obligations of the Prime Consultant and General Contractor are solely for the benefit of the Owner and are not intended in any respect to benefit any third parties.
- **18.11 Rights and Remedies.** The rights and remedies under this Agreement are the exclusive remedies available to the Parties.
- **18.12 Survival.** The following provisions will survive the termination or expiration of this Agreement: (i) Article 8, (ii) Section 10.13; (iii) Article 13; (iv) Article 15; (v) Article 16; (vi) Article 17; and (vii) Article 18.
- **18.13 Waiver.** No Party's action or failure to act will constitute a waiver of a right or duty afforded them under this Agreement, and such action or failure to act will not constitute

approval of or acquiescence in a breach of this Agreement, unless specifically agreed to in writing by the Parties.

- **18.14 Execution.** By executing this Agreement, each of the individuals represents that he or she has authority to bind the Party on whose behalf his or her execution is made.
- **18.15 Counterparts.** This Agreement may be executed in multiple counterparts, each of which will be deemed an original. When proving this Agreement, it will only be necessary to produce or account for the counterpart signed by the Party against whom enforcement is sought. Electronic copies or photocopies of this Agreement showing the true signatures of the Parties may be used for all purposes as originals.
- **18.16 Exhibits.** The Exhibits referred to in this Agreement, whether attached at the time of execution or added by Amendment signed by the Parties, are incorporated into this Agreement by reference as though set forth in full.
- **18.17 Entire Agreement**. This Agreement constitutes the entire integrated agreement between the Parties and supersedes all prior oral and written negotiations, representations, or agreements by the Parties with respect to this subject matter.

This Agreement is entered into as of the Effective Date first written above. The Corporation of the Town of Oakville: By: Name: Title: By: **Diamond and Schmitt Architects Incorporated** Name: Title: **Graham Construction and Engineering LP** Name: Title:

Integrated Project Delivery Agreement EXHIBIT A – Definitions

- 1. **"Added Value Incentive Items"** are Project elements that are added to the Base Program during the Design/Preconstruction Phase per Sections 7.2.5 and 9.8 of the Agreement.
- 2. "Adverse Weather" is high winds or unusual precipitation that prevent or substantially impedes the General Contractor's ability to perform Construction Work resulting in a delay in the Contract Time beyond the number of lost days built into the Milestone Schedule for adverse weather, as shown in Exhibit B. Work is substantially impeded if the General Contractor loses more than half of a planned and otherwise available workday except to the extent the delay is also caused by any fault, neglect, act or omission of the Prime Consultant, General Contractor, or their respective employees, subconsultant subconsultants, subcontractors or suppliers.
- 3. **"Agreement"** means this Integrated Project Delivery Agreement executed by the Parties for this Project and all of the Exhibits referenced in the Agreement.
- 4. "Allowed Claims" are defined in Section 13.2 of the Agreement.
- 5. **"Amendment"** is a document executed by the signatories to the Agreement amending the terms and/or conditions of the Agreement.
- 6. **"Applicable Law"** includes all local, provincial and federal rules, regulations, ordinances, building code or other codes, statutes, or regulations of Governmental Authorities that are relevant to proper and safe performance of the Work, including those listed in Exhibit A-1, Town Standards.
- 7. **"Base Program"** is based on the conceptual design and further defines the final quality, quantity, functionality, aesthetics, sustainability, and other requirements for the Project that are confirmed by the PMT during the Validation Phase. The Project scope includes all elements explicitly included in the Base Program and those that are reasonably necessary to accomplish the Base Program.
- 8. "Building Information Model" ("BIM") or "Model" is a parametric, computable representation of the Project design developed by the Prime Consultant, its subconsultants, and any Design-Build Subcontractors, and will include construction details developed by the Parties and their respective subconsultants and subcontractors. As used in this Agreement, references to Building Information Model, BIM, or the Model, include the primary design model or models and all linked, related, affiliated or subsidiary models developed for design, estimating, detailing, fabrication, or construction of the Project, or any portion or element of the Project. The portions of the BIM prepared by the Prime Consultant, its subconsultants and the Design-Build Subcontractors, and those portions prepared by General Contractor or Design-Assist Subcontractors under the responsible control of a licensed design professional, are Implementation Documents.
- 9. **"Business Day"** is any calendar day other than Saturdays, Sundays, and Ontario provincial holidays.
- 10. **"Business Terms Sheet"** are the page(s) under that heading after the cover page of the Agreement that sets forth the key business terms among the Parties.

- 11. "Calendar Day" or "day" is any day whether a Business Day or not.
- 12. **"Change Order Request" or "COR"** is a written request for Change Order, which sets forth the nature of the change, the reason for the change, and the effect, if any, on the Target Cost, the Contract Time, or ICL.
- 13. **"Change Order"** is a mutually agreed written order among Owner, Prime Consultant and General Contractor adjusting the Target Cost, ICL and/or Contract Time due to changes to the Base Program or Implementation Documents.
- 14. **"Chargeable Costs"** are cost categories that include the Prime Consultant and General Contractor costs incurred in the performance of the Work (excluding profit) as more specifically defined in Exhibits E and F and are chargeable against the Target Cost. Prime Consultant and General Contractor will require that profit and risk register be excluded from Chargeable Costs of their Risk/Reward Subcontractors and Risk/Reward Subconsultants.
- 15. **"Construction Phase"** is the third phase of the Project commencing on the effective date of the Notice to Proceed to construction and ending at Final Completion.
- 16. **"Construction Work"** includes all labour, materials, equipment, appurtenances and services necessary for proper construction and initial commissioning of the Project in accordance with the Contract Documents performed by General Contractor or subcontractors.
- 17. **"Contract Documents"** include this Agreement (inclusive of all Exhibits), the Building Information Model, the Implementation Documents, and all other documents issued by the Prime Consultant, its subconsultants, and Design-Build Subcontractors for construction of this Project, any PMT Bulletins, SMT Bulletins and/or Owner's Directives, and any subsequent Amendments or Change Orders. The Contract Documents include Submittals prepared by Design-Build Subcontractors and those Submittals incorporated into the Model. The documents included in the Contract Documents are complementary and what is required by one is required by all. If there are conflicting requirements within or between the various Contract Documents, the PMT will determine which requirements will better achieve the Project Objective.
- 18. **"Contract Tasks"** are those performance obligations of each Party described in the Contract Task Matrix set forth in Exhibit C.
- 19. **"Contract Time"** is the time allotted in the Milestone Schedule to achieve Final Completion of the Project.
- 20. **"Design Materials"** include the Model, Record Model, the subsidiary BIM models necessary for design and construction of the Project, all electronic design data for the Project, any related two dimensional drawings, calculations, schedules or specifications, and any other design materials, created for the Project.
- 21. "Design Services" are those professional architectural and engineering services rendered by the Prime Consultant, its subconsultants and any Design-Build Subcontractors necessary to develop and complete the Project design in accordance with the standard of care set forth in the Agreement and Applicable Law.
- 22. **"Design/Preconstruction Phase"** is the second phase of Project Work where the Owner and Risk/Reward Team engage in Target Value Design to develop Implementation Documents

within the Target Cost and obtain approval of the Implementation Documents from applicable Governmental Authorities.

- 23. "Design-Assist Subcontractors" are subcontractors whose services include participation in the design effort but that are not Design-Build Subcontractors. That participation includes provision of comments and recommendations on design elements and materials, preparation of cost opinions to inform design decisions, reviewing for constructability, trade coordination, and, where appropriate, execution of drafting efforts. Nothing in this Agreement requires the Design-Assist Subcontractors to perform any Work outside their license or contrary to the laws, codes or regulations of the Governmental Authorities.
- 24. **"Design-Build Subcontractors"** are subcontractors that provide Design Services and Design Materials required for their respective Construction Work. Design-Build Subcontractors have full architecture and engineering responsibility for their portion of the Work and will have their drawings and calculations signed and sealed by architects and/or registered professional engineers licensed in the province of Ontario.
- 25. **"Estimated Final Cost"** is the sum of incurred Chargeable Costs that has been actually incurred at the time the estimate is being made plus the estimated Chargeable Costs that will be required to complete the Project.
- 26. **"Final Actual Cost"** is the sum of all incurred Chargeable Costs upon Final Completion of the Work.
- 27. **"Final Completion of the Project"** or **"Project Final Completion"** is the Final Completion of all Project Stages and completion of all requirements of the Agreement.
- 28. **"Final Completion"** occurs on the date when all of the following have occurred: General Contractor has completed the Construction Work in full compliance with the Implementation Documents; all Final Punch List items have been completed and accepted by the PMT; the Project has been commissioned; all close-out documentation required under the Contract Documents has been transmitted to Owner; a final certificate of occupancy has been issued by the Governmental Authority having jurisdiction over occupancy of the Project; and the PMT has issued a certificate of Final Completion. A Project Stage may have its own Final Completion date, but all Project Stages must be complete before Project Final Completion.
- 29. **"Final Payment"** is Owner's payment of all amounts due and owing to the Prime Consultant and General Contractor, including any ICL due after Final Completion of the Project, upon receipt of all required unconditional waivers and releases and close-out documents.
- 30. **"Final Punch List"** is the punch list prepared after Substantial Completion and final inspections documenting all Construction Work that needs to be corrected or completed to achieve Final Completion.

- 31. **"Fire Hall"** means the two storey fire hall to be constructed at the corner of Kerr and Rebecca Streets as part of the Project.
- 32. **"Force Majeure"** means natural disasters; named storms; labour strikes that cannot be resolved through a dual gate or other measures; disruptions in utility service and/or connections not caused by the General Contractor or those for which it is responsible; Governmental Authority actions other than permitting, design review or inspection of construction; civil disobedience, an act of terror, unavoidable casualties or catastrophic events; provided the above events are beyond the Prime Consultant's and/or General Contractor's control, and not due to any act or omission of the Prime Consultant and/or General Contractor or anyone for which either is responsible.
- 33. "General Contractor" means [General Contractor].
- 34. **"Governmental Authority" or "Governmental Authorities"** includes all federal, provincial, regional, municipal, district or quasi-governmental boards, departments, courts, offices or agencies that have jurisdiction over the Project.
- **35.** "Hazardous Materials" means any and all pollutants, wastes, flammables, explosives, radioactive materials, hazardous or toxic materials, hazardous or toxic wastes, hazardous or toxic substance or contaminant and all other materials governed by OSHA or other Applicable Law for environmental protection, or any substance or material that has been determined, or during the time of performance of the Contract Work is determined, to be capable of posing a risk of injury to health, safety, property or the environment by any Governmental Authority.
- 36. **"ICL Percentages"** are the percentage shares in the ICL of the individual members that make up the Risk/Reward Team and will be used for distribution of any ICL.
- 37. "Implementation Documents" consist of the Model; plans, sections and elevations extracted from the Model; and any ancillary drawings, specifications and construction details together with dimensions and layouts for civil, architectural, structural, mechanical, electrical, plumbing systems, and landscape design. The Implementation Documents will describe in detail the requirements for the Construction Work and provide information necessary and appropriate to obtain all necessary permits for construction of the Project.
- 38. "Incentive Compensation Layer" ("ICL") is an amount that is increased or decreased based on Project outcome during the Project and is distributed to the Risk/Reward Team in accordance with their ICL Percentages and the ICL Distribution Spreadsheet (Exhibit G). Payment of the ICL is contingent upon the Project phasing milestone disbursement conditions set forth in the Agreement.
- 39. "Integrated Project Delivery Team" ("IPD Team") and "IPD Team Member" are defined in Section 3.1.
- 40. **"Joint Site Investigation"** is a site investigation attended by the Risk/Reward Team during the Validation Phase for the purpose of reviewing existing information and investigating the Project site to identify deficiencies and discrepancies, and to determine the extent of any additional investigations or testing required for proper design and construction.

- 41. **"Milestone Schedule"** is a Project schedule submitted during the Validation Phase that projects Milestones for design and construction, including anticipated completion of design and procurement of permit(s), procurement of long lead items, projections for key Construction Work, Owner's occupancy requirements, and projections for Substantial Completion and Final Completion. The Milestone Schedule will be used as a management tool only, however the Contract Time will be tied to the milestone Project Final Completion date, subject to adjustment through approved Change Order.
- 42. **"Milestone"** or **"Milestones"** are events noted in the Milestone Schedule where the Project is evaluated and, if pre-conditions are met, a portion of ICL is distributed to the Prime Consultant and General Contractor.
- 43. **"Notice to Proceed"** is a written document issued by the Owner or the PMT to set the date of commencement of the Construction Phase of the Project or for various Project Stages.
- 44. "Owner" means The Corporation of the Town of Oakville.
- 45. **"Owner's Budget"** means the maximum amount which the Owner has budgeted for the payment of Chargeable Costs and ICL with respect to the Project. Subject to amendment by the Owner, the Owner's Budget is [\$33.2 million].
- 46. **"Owner's Directive"** is a written directive from the Owner that overrides a decision by PMT or the SMT. An Owner's Directive may be construed as an Owner-Elected Change if it affects the Target Cost and/or Contract Time.
- 47. **"Owner-Elected Change"** is a change to the scope of the Work described in the Base Program or Implementation Documents that either impacts the Target Cost and is not reasonably inferred from the Base Program or Implementation Documents, or required as a result of either coordination of the Construction Work or design error or omission.
- 48. **"Permitted Delay"** means a critical path delay meeting one of the categories described in Section 10.5 of the Agreement.
- 49. **"Party"** means the Owner, Prime Consultant, or General Contractor and **"Parties"** means all of them.
- 50. **"Phase"** is a functional segregation of Project Work into Validation, Design/Preconstruction and Construction. Each Project Stage may have its own Phases.
- 51. **"PMT Bulletin"** is a written directive from the Project Management Team derived from a unanimous vote that affects design, cost, schedule, or allocation of the Work. A PMT Bulletin may affect the Project Objective.
- 52. **"Post Permit Change"** is a substantive change in the permit by a Governmental Authority or as a result of changes to laws or regulations that impact the Construction Work subsequent to the issuance of permit provided that the changes are not due to coordination of the Construction Work or design errors and omissions and provided that the changes were not reasonably known or anticipated when the Target Cost was set.
- 53. "Prime Consultant" means [Prime Consultant].
- 54. "Product Data" are illustrations, standard schedules, performance charts, instructions,

brochures, diagrams and other information furnished by the General Contractor or a subcontractor, tier- subcontractor, manufacturer, supplier or distributor to illustrate materials or equipment for some portion of the Construction Work.

- 55. **"Project"** means the project described in Section 2.1 of the Agreement, and includes the Validation Phases, Design/Preconstruction Phases, and Construction Phases for all Project Stages.
- 56. **"Project Implementation Teams" ("PITS")** are interdisciplinary groups of Project Participants organized by the PMT. PITS are part of the collaboration process to develop the Implementation Documents and other deliverables, and may be formed temporarily or for the duration of the Project.
- 57. **"Project Management Team" ("PMT")** consists of representatives from the Owner, Prime Consultant, and General Contractor that will act in a collaborative manner to provide management level leadership during the design and construction process in a concerted effort to achieve the Project Objective.
- 58. **"Project Objective"** includes all Owner requirements, goals and limitations, is comprised of the Base Program, Added Value Incentives Items, and Implementation Documents, and includes the Target Cost and Contract Time. The Project Objective establishes all requirements for measuring the Project's success.
- 59. **"Project Participant"** is any person or entity that is providing material, equipment, work or services for the Project.
- 60. **"Project Stage"** is a portion of the Project that is geographically or otherwise distinct and may have separate Milestones and dates for Substantial and Final Completion.
- 61. **"Record Model"** is the version of the BIM that will be updated throughout construction to reflect the as-built condition of the Project and is turned over to the Owner upon Final Completion.
- 62. "Regular Meeting" is defined in Section 5.1.4(a).
- 63. "Request for Information" ("RFI") is defined in Section 5.1.3.
- 64. "Risk/Reward Subconsultants" are subconsultants engaged by the Prime Consultant to perform Design Services per the obligations stated in their respective consulting agreement and that have placed 100% of their profit at risk and are therefore eligible to share in the incentives provided and risk associated with the Project's success.
- 65. "Risk/Reward Subcontractors" are subcontractors and Vendors engaged by the General Contractor for performance of a portion of the Construction Work per the obligations stated in their respective subcontract agreement or purchase order and that have placed 100% of their profit at risk and are therefore eligible to share in the incentives provided and risk associated with the Project's success.
- 66. "Risk/Reward Team" consists of the Prime Consultant, General Contractor and all of their respective Risk/Reward Subconsultants and Risk/Reward Subcontractors, each of whom have placed their respective profit at risk, are eligible to share in the ICL per the Agreement, and are included in the mutual waiver of liability provisions set forth in the Agreement.
- 67. "Samples" are physical examples that illustrate materials, equipment or workmanship and

establish standards by which the Construction Work will be judged.

- 68. **"Senior Management Team" ("SMT")** includes a senior executive member from the Owner, Prime Consultant, and General Contractor that will act collaboratively to resolve any matters referred to it by the PMT by a majority vote, subject to an Owner's Directive.
- 69. **"Separate Contractors"** includes subconsultants, contractors and vendors procured directly by the Owner to perform work or services on the Project.
- 70. **"Set Based Design"** is a design strategy that advances in parallel alternative design solutions that meet Project criteria and constraints until a decision is made to select one solution over the alternatives.
- 71. **"Shop Drawings"** are drawings, diagrams, schedules and other data specially prepared for the Construction Work by the General Contractor or a subcontractor, tier-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- 72. **"SMT Bulletin"** is a written directive from the SMT derived from a majority vote of the SMT and is binding on all Project Participants unless vetoed or modified by an Owner's Directive.
- 73. **"Special Meeting"** is defined in Section 5.1.4(b).
- 74. **"Staging Schedule"** is used if the Project will be performed in stages. At a minimum, the Staging Schedule defines the dates for commencement of construction, Substantial Completion, and Final Completion of each Project Stage.
- 75. **"Standard Consultant"** is a subconsultant engaged by the Prime Consultant that has not placed profit at risk and therefore is not eligible to share in the Agreement's financial incentives, ICL and mutual liability waivers. Standard Consultants are not a part of the Risk/Reward Team.
- 76. **"Standard Subcontractor"** is a subcontractor or supplier engaged by the General Contractor that has not placed profit at risk and therefore is not eligible to share in the Agreement's financial incentives, ICL and mutual liability waivers. Standard Subcontractors are not a part of the Risk/Reward Team.
- 77. **"Submittals"** include Shop Drawings, Product Data and Samples, but are not Contract Documents unless they are produced and stamped by a Design-Build Subcontractor. To the extent required by the Contract Documents, all Submittals that are not produced by a Design-Build Subcontractor only demonstrate how the General Contractor and subcontractors propose to execute the Construction Work shown by the Contract Documents.
- 78. **"Substantial Completion"** occurs on the date when the Project is sufficiently complete to allow the Owner to legally occupy or utilize the Project for its intended purpose, all systems are operational as designed or required, all inspections and tests required under the Contract Documents have been completed successfully, the Governmental Authority with jurisdiction over Project occupancy has issued approval for legal occupancy, and the PMT has issued a certificate of Substantial Completion. A Project Stage may have its own Substantial Completion date.
- 79. **"Target Cost"** is the initial cost of the Work established during the Validation Phase and Target Value Design by the Owner and Risk/Reward Team members which includes all Chargeable Costs to complete the Work described in the Base Program. The lower of (i) the Owner's Budget and (ii) the Target Cost will be used as the threshold for which the Added Value

Incentive Items and Base Program are compared in order to determine whether ICL is increased during the Design/Preconstruction Phase.

- 80. **"Target Value Design"** is a design discipline that requires project values, cost, schedule, and constructability to be basic components of the design criteria, and uses cost targets to drive innovation in designing a project to provide optimum value to an owner. Target Value Design uses constructability and cost information from the Risk/Reward Team before design decisions are made to allow the design to progress within the Target Cost and Contract Time.
- 81. "Unforeseen and Differing Site Conditions" means discovery of an unknown, subsurface or otherwise concealed physical condition that differs materially from those indicated in the Implementation Documents or the information obtained from the Joint Site Investigation; an unknown physical condition of an unusual nature that differs materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character and nature provided for in the Implementation Documents; or an unknown, pre-existing hazardous substance or condition that requires removal or remediation.
- 82. **"Validation Phase"** is the first phase of the Project where the Owner and Risk/Reward Team members document the Base Program and establish the Target Cost and the Added Value Incentive Items.
- 83. **"Vendors"** are suppliers that are engaged directly by the General Contractor that have placed their profit at risk and are participating on the Risk/Reward Team.
- 84. "Willful Default" is one of the following events: (i) actual or constructive abandonment of the Project; (ii) persistent and repeated failure, after written notification, to correct Work that significantly and materially deviates from the Implementation Documents or applicable laws, codes or regulations; or (iii) fraud, reckless disregard or willful misconduct. Actual abandonment occurs if the Party, without justification, ceases performing Work for a period of 21 consecutive days or notifies one of the Parties that it will not be performing Work on the Project anymore. Constructive abandonment occurs if the Party, without justification, expends so little effort on the Project that there is no meaningful progress for 21 consecutive days. The good faith exercise of any contractual suspension rights granted the Risk/Reward Team member under this Agreement or an applicable subcontract or consulting agreement is not an intentional or constructive abandonment.
- 85. **"Work"** includes all labour, materials, equipment, appurtenances and services required to design, construct, and commission the Project in accordance with the Contract Documents.

INTEGRATED PROJECT DELIVERY AGREEMENT

EXHIBIT A-1 – TOWN STANDARDS

1. Town of Oakville and Accessibility:

The Town is committed to providing accessible facilities, programs and services and to be the most livable town in Canada. The **Oakville Universal Design Standards** (*OUDS*) was developed to help staff realize this commitment and provides an innovative and detailed approach for the design of barrier free and accessible facilities. The design and construction of Town buildings must meet all of the requirements under OUDS and its use is mandatory for all construction projects at Town owned or leased facilities including new construction, additions, renovations and capital replacements.

Refer to the Town's website to download a copy of the *OUDS* at: http://www.oakville.ca/assets/general%20-%20residents/auniversal-design-standards.pdf

2. Sustainable Design and Construction at the Town of Oakville:

The Town is committed to providing programs and services that are environmentally sustainable. The Town's Official Plan - Livable Oakville - sets out clear objectives with respect to the environment and seeks to reduce the overall impact of the built environment on human health and the natural environment by:

- a) minimizing the Town's ecological footprint
- b) achieving sustainable building and community design; and
- c) enhancing the Town's air and water quality.

In order to meet this mandate, the design and construction of Town buildings must employ sustainable means and methods with a particular focus on the appropriate and efficient use of resources (energy, water and materials) towards reducing the building's environmental impact during its lifecycle. In keeping with Town policy, the Project must achieve minimum LEED Silver certification and both buildings must meet high performing building criteria established by ASHRAE Standard 189.1 2014.

3. CADD Drawing Standard and BIM:

The Town currently uses and supports AutoCAD V 12 and all Project drawings developed by the team must be compatible with this version. As well, the Project will be utilizing BIM as a design and construction tool towards early and coordinated information and documentation to help the team better visualize and analyze performance, appearance and cost. One of the first tasks of the Project will be establishing BIM protocols using the Town's AutoCAD standard as a starting point.

4. Accommodation and Furniture Standard:

The Town has a corporate Accommodation and Furniture Standard in place in order to provide a consistent approach to cost, quality, and health and safety issues and its use is mandatory for all Town projects. The standard defines the physical components of individual work setting spaces and teaming spaces, a universal space standard (an area per person benchmark for high level

planning) and optimal workstation and office layouts based on functionality rather than hierarchy. The design and install of furniture components will be managed by the Facilities and Construction Management department and the Design Team will need to coordinate with Town staff as well as Workplace Resource Canada - Herman Miller (vendor of record).

5. Corporate identity Standards:

All signage and logos are to be designed in accordance with the town's corporate identity standards. In addition, the town has developed a Pylon Sign Standard and Interior Signage Standards for town facilities to be met by the Design Team.

6. Communications and Security Systems Standards:

The Town of Oakville's Data Centers, Network, Telephony and Life Safety systems follow industry standards as specified in TIA-942 Specifications for Data Center Infrastructure, ASHREA, Uptime Institute, PCI-DSS, ISO, SSAE-16 and all Ontario Provincial Standards. Sustaining and ancillary equipment along with life safety and security utilize leading edge technology are provided from partners such as Mitel Phone Systems, Schneider Electric and APC UPSs for power conditioning and distribution in computer rooms and wiring closets, Systemax for UTP Copper and Fiber, CORNING for Fiber, INERGEN for Computer Room Fire Suppression, Canseco Systems for Door Access and Biometric Time and Attendance and Panasonic for CCTV Systems. The Town ensures optimal network efficiency and availability by deploying redundancy and utilizing standard methodologies defined by the appropriate governing institution.

Integrated Project Delivery Agreement EXHIBIT B-1 – Owner's Program Requirements

- **1. Owner's Program Requirements.** The Owner's Program Requirements that are reflected in the Target Cost (Exhibit B-2) consists of:
 - **1.1** The standards and quality requirements of the IPD Agreement;
 - **1.2** The information in the following documents:
 - 1.2.1 [insert].
 - **1.3** Scope that is reasonably inferable from the above documents; and
 - **1.4** Requirements of all applicable Governmental Authorities.
- **2. Specific Inclusions and Exclusions.** The following are specifically excluded from the Owner's Program Requirements except for partial inclusions as noted.
 - 2.1.1 [insert]
- 3. Attachments
 - 3.1.1 None

End of Exhibit

Integrated Project Delivery Agreement EXHIBIT B-1a – Fire Hall Building Program Summary

Integrated Project Delivery Agreement EXHIBIT B-1b – Oakville Arena Building Program Summary

Integrated Project Delivery Agreement EXHIBIT B-2 – Target Cost Breakdown

TOTAL	
DELTA FROM RES. ENVELOPE	
DELTA FROM RES. INTERIOR	
TOTAL	
RISKS & INNOVATIONS	
SUBTOTAL	
Contingency used	
TOTAL	

Integrated Project Delivery Agreement EXHIBIT B-2a - Risks and Innovations (as of [date])

Integrated Project Delivery Agreement EXHIBIT B-4 – Milestones

Milestone #	Milestone	Condition of Satisfaction
1	Validation	Agreement on Owner's Program Requirements, Agreement on Initial Project Schedule, Agreement on Target Cost, including Target Cost Breakdown and documentation.
2	Foundation start	
3	Building enclosed • Fire Hall • Oakville Arena	Roof, Exterior Walls to AVB with Windows installed Roof, Exterior Walls to AVB with Windows installed
4	Substantial Completion	Building is considered substantially complete as defined by Ontario Construction Lien Act Building is considered substantially complete as defined by Ontario Construction Lien Act
5	Final Completion	When Project is considered Final Completion as defined by the Ontario Construction Lien Act.

End of Exhibit

Integrated Project Delivery Agreement EXHIBIT D – Intentionally Omitted

Integrated Project Delivery Agreement EXHIBIT E – Prime Consultant's Chargeable Costs

The Chargeable Costs of Prime Consultant (and its Risk/Reward Subconsultants) include only those actual costs (less discounts and rebates) reasonably and necessarily incurred in the performance of their respective obligations under the IPD Agreement and meeting one of the categories set forth in Sections 1 or 2 below, and which in any case are not excluded under Section 3. All expenses from Section 1 through 4 are without overhead and mark-up. All other costs are excluded unless approved in advance by the PMT.

1. Chargeable Costs

- **1.1 Prime Consultant Design Services**. The Chargeable Cost for Prime Consultant's employees will be charged based on the hourly rates using a 2080 hour work year and an hourly multiplier agreed with Owner.
 - (a) The hourly rates must be computed using the employee's direct salary expense; the amount paid to the employee as wages, excluding benefits (health insurance, long term disability, sick leave, pension, and vacation accruals), any profit sharing plans, bonuses or other discretionary labour charges, and profit.
 - (b) The hourly rate multiplier is made up of: vacation, sick days, employment insurance (EI), Canada Pension Plan (CPP), WSIB, Long Term Disability (LTD), health plan, sick leave, pension/RRSP and bonuses. The multiplier does not include: profit sharing plans, other corporate expenses, training except specific to the Project, administration, promotion, general office time not applied to the Project and Miscellaneous Expenses in section 2. The hourly rate multiplier does not include any amount of profit.
 - (c) The chargeable rate for non-listed employees will be calculated based on the employee's direct salary expense multiplied by the agreed multiplier set forth in the Business Terms Sheet and deriving the hourly rate using a 2080 hour work year. The hourly rates may not include overhead (beyond the multiplier), profit, or paid time-off.
- **1.2** Risk/Reward Subconsultants. All Risk/Reward Subconsultant agreements must include a Chargeable Cost section consistent with this Exhibit E that specifically defines the categories of labour, the direct salary expense for those employees, and billable rates based on an agreed multiplier that does not include profit or paid time-off.
- **1.3 Limitation on Cost of Salaried Personnel**. Charges for Work performed by salaried personnel of Prime Consultant or its Risk/Reward Subconsultants in excess of the 170 hour work month are not allowable, unless agreed to by the PMT prior to the time being incurred and only in an amount agreed to by the PMT as appropriate.
- **1.4 Fixed Price Consultants**. Payments made by Prime Consultant to fixed price subconsultants per their written consulting agreements and the Agreement.

2. Miscellaneous Expenses

Miscellaneous Expenses that are an allowable cost to the project. Miscellaneous Expenses are accrued without overhead and profit.

- **2.1 Services**. Actual costs for teleconferencing, video-conferencing, and express delivery or messenger services or other services purchased on behalf of the Project and with prior approval of PMT.
- **2.2 Travel Expenses**. All reasonably and actually incurred direct, non-salary, travel-related reimbursable expenses required for the Project (not including meals) will be billed to Owner at actual cost without markup. Mileage will be reimbursed at [\$0.54] per kilometer for travel by Prime Consultant in its own vehicles and based on distance from office to Big Room and/or site.. All air travel, regardless of domestic or international destination, will be at unrestricted coach class fare or other class, whichever is lowest. All air travel cost must be approved by PMT prior to travel.

- **2.3 Models & Mock-Ups**. Actual cost of physical models and mock-ups requested by PMT.
- **2.4** Fees. License fees paid for the use of a particular design or process required by the IPD Agreement and approved by the PMT. Filing or other fees paid for securing approval of Governmental Authorities for the design of the Project.
- **2.5 Insurance**. Insurance expenses allocable to this Project to the extent specifically permitted in the IPD Agreement. Deductibles are a Chargeable Cost only to the extent specifically provided in Exhibit H.
- **2.6 Reproduction**. The cost of reproduction and printing of Implementation Documents will be based on the following rates:

8x10 Color Laser Printing/Copies	
11x17 Color Laser Printing/Copies	
Plotting	
Color Plotting	
Half Size Printing/Reproduction (15x21)	
Full Size Printing/Reproduction (30x42)	

- **2.7 Taxes**. Sales, use, gross receipt, HST, or similar taxes imposed by a Governmental Authority that are related to the Prime Consultant's services, excluding any franchise or income based tax.
- **3. Excluded Costs.** The following costs are not Chargeable Costs and therefore are non-reimbursable expenses:
- **3.1** Salaries and other compensation of the Prime Consultant's personnel stationed at any principal office or offices other than the site office, except as specifically included in Section 1.1.
- **3.2** Employee bonuses or incentive program payments regardless of whether personnel are specifically assigned to this Project. This cost is carried in the Prime Consultant's and their Risk/Reward Subconsultant's respective multipliers.
- **3.3** Expenses related to the firm's principal office and offices other than the Project site office because this cost is carried in the Prime Consultant's and their Risk/Reward Subconsultant's respective multipliers.
 - **3.4** Any cost resulting from fraud, willful default or willful misconduct.
- **3.5** Any cost resulting from negligence once the risk register within the Target Cost is exhausted.
 - **3.6** Costs incurred after Project Final Completion.

4. Overhead.

Home office overhead costs allocable to this Project in accordance with Generally Accepted Accounting Practices are charged through the agreed overhead rate set forth in Section 7.1.1 of the IPD Agreement. The agreed overhead rate compensates the Prime Consultant for costs or expenses it incurs other than the Chargeable Costs set forth in Sections 1 and 2 and not excluded by Section 3 of this Exhibit. The overhead rate may not include any amount of profit.

Page left blank

Integrated Project Delivery Agreement EXHIBIT F – General Contractor's Chargeable Costs

The Chargeable Costs of General Contractor (and its Risk/Reward Subcontractors) include only the actual costs (less discounts and rebates) reasonably and necessarily incurred in the performance of their respective obligations under the IPD Agreement and meeting one of the categories set forth in Sections 1 through 4, inclusive, and which in any case are not excluded under Section 6. All expenses from Section 1 through 4 are without overhead and profit. All other costs are excluded unless approved in advance by the PMT.

1. Chargeable Costs

- **1.1 General Contractor's Personnel**. Wages and salaries of General Contractor's supervisory and administrative personnel assigned to the Project (other than field labour) to perform project management, superintending, preconstruction services, engineering, design, planning and scheduling, safety, purchasing, estimating, IT, and other supervisory activities as preapproved by the PMT, whether performed at the site or off-site offices. Personnel may include accounting and administrative services performed off-site if pre-approved by the PMT. The percentage of time assigned to this Project and the agreed hourly rates charged for such personnel will be agreed to by PMT. The percentage of time may be adjusted by the PMT. The Chargeable Cost for General Contractor Personnel will be charged based on the hourly rates using a 2080 hour work year and an hourly multiplier agreed to by the Owner.
 - (a) The hourly rates must be computed using the employee's direct salary expense; the amount paid to the employee as wages, excluding benefits (health insurance, long term disability, sick leave, pension and vacation accruals), any profit sharing plans, deferred compensation and profit.
 - (b) The hourly multiplier is made up of vacation, sick days, employment insurance (EI), Canada Pension Plan (CPP), WSIB, Long Term Disability (LTD), health plan, sick leave, pension/RRSP, cell phone expense, deferred compensation, supplemental dues (if applicable), training directly related to the Project (or required by union contract), any other benefits or payments required by law or General Contractor and/or subcontractor collective bargaining agreements. The multiplier does not include: profit sharing plans, other corporate expenses, training except specific to the Project, administration, promotion, general office time not applied to the Project and Miscellaneous Expenses in section 2. The hourly rate multiplier does not include any amount of profit.
 - (c) The hourly multiplier does not include: profit sharing plans, other corporate expenses, training except specific to the Project, administration, promotion, general office time not applied to the Project and Miscellaneous Expenses listed in Section 4. The hourly rate multiplier may not include overhead and profit.
 - (d) The chargeable rate for non-listed employees will be calculated based on the employee's direct salary expense multiplied by the agreed multiplier set forth in the Business Terms Sheet and deriving the hourly rate using a 2080 hour work year. The hourly rates may not include overhead (beyond the multiplier), profit, or paid time-off.
- **1.2 Field Labour**. Field labour includes wages of construction workers directly employed by General Contractor or Risk/Reward Subcontractor to perform construction work on-site. Field labour chargeable costs will be determined based on Section 1.1 above. Off-site prefabrication and other off-site field labour are addressed in Section 1.4.

- **1.3 Subcontractor Costs**. Payments made by General Contractor to subcontractors in accordance with the requirements of their written subcontract agreements and the IPD Agreement. All contracts with Risk/Reward Subcontractors must include a Chargeable Cost section consistent with this Exhibit F that specifically defines the categories of labour that are reimbursable and the labour rates in dollar amount per hour, including any overtime rates for field labour.
- 1.4 Off-Site Shop Labour Costs. Pre-fabrication of components and assemblies is a key strategy for improving quality, reducing cost, expediting the schedule and improving worker safety. However, monitoring costs for off-site labour is challenging because the labour may be incurred without PMT representatives being present. In order to maintain reasonable financial oversight without discouraging off-site fabrication, the General Contractor or its subcontractors will prepare resource loaded work plans for their off-site labour that will be reviewed and if acceptable, approved by the PMT. The hourly rates for off-site shop labour will be agreed prior to off-site work commencing, will not include profit (if the provider is a Risk/Reward Subcontractor), and will be based on agreed rates using a methodology similar to the standards used for field labour, but subject to appropriate modifications for shop overhead. Incurred off-site shop labour that is consistent with the submitted resource loaded work plan and charged at the agreed rates, is payable as a Chargeable Cost.

2. Materials and Equipment Incorporated into the Project

- **2.1 Materials and Equipment**. Direct costs, including transportation and storage, of materials and equipment incorporated or to be incorporated into the Project. Purchases or costs from any affiliated entities, divisions, groups, etc. for materials, equipment, and other costs will be billed at the actual and auditable cost incurred by those affiliated entities, divisions, groups, etc., except as pre- approved by the PMT.
- **2.2 Waste and Spoilage**. Costs of materials described in Section 2.1 in excess of those actually installed to allow for reasonable waste and spoilage. Unused excess materials, if any, will become Owner's property at final completion or, at Owner's option, will be sold by the General Contractor and credited against the final cost of the Project.

3. General Requirements

- 3.1 Temporary Facilities & Services. Costs for General Contractor's on-site office or big room and temporary facilities (including trailers, power, water, sanitary, utilities, gas, telephone, internet), security, temporary and winter protection and barricades (including fences, tarps, signage and traffic control), temporary office furniture and equipment (including cost of computers and software purchased for this Project with the concurrence of the PMT and inclusive of all variable software, applications, systems, and support costs directly related to this Project), devices, servers, printers, copiers, plotters, facsimile transmissions, long-distance telephone calls, postage and parcel delivery charges, food for Big Room and as approved by PMT, etc.; including costs of transportation, installation, minor repairs and replacements, dismantling and removal thereof. Costs will be billed at the actual and auditable cost incurred. Any equipment or materials purchased and charged to the Project will remain the property of Owner at the end of its usage.
- **3.2** Temporary Materials and Equipment. Costs, including transportation and storage, installation, maintenance, dismantling and removal of construction materials, supplies, machinery, equipment, and hand tools not customarily owned by the construction workers, that are provided by the General Contractor at the site (but not incorporated into the Project) and fully consumed in the performance of the construction work. If items are not fully consumed, the cost less salvage value, whether sold to others or retained by the General Contractor, will be included. Salvage value will be the fair market value or actual value received. All charges for materials equipment and facilities charged to the Project are to be at the General Contractor's actual and auditable costs included in Exhibit F-2.
- 3.3 Rental Expenses. Rental charges for non-affiliated third-party equipment will be considered reimbursable and will be reimbursed at actual costs, as long as rental rates are consistent with those prevailing in the locality. Such equipment will be exclusive of small tools, and limited to the direct costs of transportation, delivery, installation, dismantling, removal, and maintenance. Rental charges will be equitably prorated if the foregoing equipment is not exclusively devoted to the Project. For rental charges for items described in Section 3.2 that are rented from the General Contractor or an affiliate, the aggregate rental amounts (exclusive of all installation, maintenance, dismantling, removal, transportation, and delivery costs) for any one piece of equipment cannot exceed 80% of the purchase price (at the time it is placed in service) during the rental period of the equipment used for this Project. Agreed rates for equipment that is owned by the General Contractor or an affiliate and rented to the Project are subject to these same terms and are included in Exhibit F-2. Upon Owner's request, General Contractor will present an analysis of an opportunity to purchase rather than rent the item. All purchased items will be a Chargeable Cost and title to the property will vest to Owner upon Final Completion.

- **3.4 Materials & Equipment Stored Off-Site**. Costs of materials and equipment suitably stored off-site at a mutually acceptable location, if pre-approved by the PMT and stored in accordance with the Agreement. All costs are to be at the General Contractor's actual and auditable costs.
- **4. Miscellaneous Expenses.** Miscellaneous Expenses that are a chargeable cost. Miscellaneous Expenses are accrued at cost without overhead and profit.
- **4.1 Taxes**. Sales, use, gross receipt, or similar taxes imposed by a Governmental Authority that are related to the Construction Work, excluding any franchise or income based tax.
- **4.2 Insurance**. Insurance expenses specifically required by the IPD Agreement and allocable to this Project. Deductibles are a Chargeable Cost only to the extent specifically provided in Exhibit H.
- **4.3 Permits, Fees, and Assessments**. Fees and assessments for the building permit and for other permits, licenses and inspections that General Contractor is required to pay under the IPD Agreement. Major permits may be paid by Owner directly.
 - **4.4 Testing**. Direct fees of laboratories for tests required by the IPD Agreement.
- **4.5** Fees. License fees paid for the use of a particular design or process required by the IPD Agreement and approved by the PMT.
- **4.6 Recycle & Waste**. Direct costs of removal and disposal of debris from the site and recycle costs not offset by recycle fees or rebates.
- **4.7 Document Reproduction**. Direct costs for blueprinting and other document reproduction necessary for constructing and administrating the Project.
- **4.8 Travel Expenses**. All reasonably and actually incurred direct, non-salary, travel-related reimbursable expenses necessary to the Project (not including meals) will be billed to Owner at actual cost without markup. Mileage will be reimbursed at [\$0.54] per kilometer for travel by General Contractor in its own vehicles and based on distance from office to Big Room and/or site. All air travel, regardless of domestic or international destination, will be at unrestricted coach class fare or other class, whichever is lowest. All air travel cost must be approved by PMT prior to travel.
- **4.9 Emergencies & Repairs**. Subject to the IPD Agreement, costs incurred in taking action to prevent threatened damage, injury, or loss in case of an emergency that threatens the safety of persons.
- **4.10 Trade Discounts and Surplus**. Trade discounts and refunds for General Contractor purchased material and equipment will be credited against the incurred Chargeable Cost. Sales of surplus materials and equipment will likewise be credited against the Chargeable Costs.

5. Overhead.

Home office overhead costs allocable to this Project in accordance with Generally Accepted Accounting Practices are charged through the agreed overhead rate set forth in Section 7.1.1 of the

IPD Agreement. The agreed overhead rate compensates General Contractor for all costs or expenses it incurs, of any nature, other than the Chargeable Costs set forth in Sections 1 through 4 of this Exhibit. The overhead rate may not include any amount of profit. The overhead rate will be applied against all Chargeable Costs.

- **6. Excluded Costs.** The following costs are not Chargeable Costs and therefore are non-reimbursable expenses.
- **6.1** Salaries and other compensation of the General Contractor's personnel stationed at any principal office or offices other than the site office, except as specifically included in Section 1.2.
- **6.2** Expenses of the General Contractor's or its Risk/Reward Subcontractors' principal offices, and offices other than the Project site office.
- **6.3** All costs of business and/or operating permits, licenses, fees and taxes, required by any Governmental Authorities or labour agreements to enable the General Contractor or its subcontractors of any tier to be qualified to do business and/or perform trade activities and/or any Work.
 - **6.4** Any cost resulting from fraud, willful default or willful misconduct.
- **6.5** Any cost resulting from negligence once the risk register within the Target Cost is exhausted.
 - **6.6** General Contractor's capital expenses, including interest on its capital employed for the Work.
 - **6.7** Rental costs of machinery and equipment, except as specifically provided in Section 3.3
 - **6.8** Costs incurred after Project Final Completion.
- **7. Taxes**. Taxes imposed by governmental authority that are related to the Project will be against the total amount due as part of the progress billing.

Page left Blank

Integrated Project Delivery Agreement EXHIBIT F-1 - General Contractors' Chargeable Rates

Integrated Project Delivery Agreement EXHIBIT G - ICL Distribution Spreadsheet

Integrated Project Delivery Agreement EXHIBIT H - Insurance Summary

Integrated Project Delivery Agreement EXHIBIT I – Key Employees